

Digital Object Identifier 10.1109/OJVT.2022.3174546

# **In-Vehicle Sensing for Smart Cars**

XIAOLU ZENG <sup>1</sup>, FENGYU WANG <sup>2</sup> (Member, IEEE), BEIBEI WANG <sup>3</sup> (Senior Member, IEEE), CHENSHU WU<sup>03,4</sup> (Senior Member, IEEE), K. J. RAY LIU<sup>03</sup> (Fellow, IEEE), AND OSCAR C. AU<sup>3</sup> (Fellow, IEEE)

(Invited Paper)

<sup>1</sup>School of Information and Electronics, Beijing Institute of Technology, Beijing 100811, China <sup>2</sup>School of Artificial Intelligence, Beijing University of Posts and Telecommunications, Beijing 100876, China <sup>3</sup>Origin Wireless Inc., Greenbelt, MD 20770 USA <sup>4</sup>Department of Computer Science, The University of Hong Kong, Hong Kong, China CORRESPONDING AUTHOR: XIAOLU ZENG (e-mail: feiying3709@sina.com)

**ABSTRACT** Driving safety has been attracting more and more interest due to the unprecedented proliferation of vehicles and the subsequent increase of traffic accidents. As such the research community has been actively seeking solutions that can make vehicles more intelligent and thus improve driving safety in everyday life. Among all the existing approaches, in-vehicle sensing has become a great preference by monitoring the driver's health, emotion, attention, etc., which can offer rich information to the advanced driving assistant systems (ADAS) to respond accordingly and thus reduce injuries as much/early as possible. There have been many significant developments in the past few years on in-vehicle sensing. The goal of this paper is to provide a comprehensive review of the motivation, applications, state-of-the-art developments, and possible future interests in this research area. According to the application scenarios, we group the existing works into five categories, including occupancy detection, fatigue/drowsiness detection, distraction detection, driver authentication, and vital sign monitoring, review the fundamental techniques adopted, and present their limitations for further improvement. Finally, we discuss several future trends for enhancing current capabilities and enabling new opportunities for in-vehicle sensing.

**INDEX TERMS** Artificial intelligence, advanced driving assistant systems (ADAS), distraction/inattention, driver authentication, fatigue/drowsiness, in-vehicle sensing survey, occupancy detection, smart car, vital sign monitoring, wireless sensing.

#### I. INTRODUCTION

The last several decades have witnessed the unprecedented proliferation of automobiles, which has contributed greatly in our daily commute, economy, business and entertainment [1]. According to the American Automobile Association (AAA) [2], there are roughly about 1.2 billion vehicles operating on the planet every day with an average trip of 15 minutes. The in-vehicle time grows up to 46 minutes per day in the United States [3]. While we have benefited a lot from the tremendous number of motor vehicles, it has been shown [4] that the road accidents cause approximately 1.3 million deaths every year and about 20-50 million more non-fatal injuries, many of which incur a lifelong disability [5]. Among those accidents, about 94% -96% of them are related to some human error [6].

To improve driving safety, many efforts have been devoted by both the government and car manufacturers such as

legislatively prohibiting the use of wireless devices and disabling some of the amusement features (i.e., Bluetooth setting) during driving. However, driving is a complex task and requires a combination of cognitive engagement and physical operations, which makes it very hard for a driver to concentrate, especially during long trips such as those for truck drivers.

As a promising safety enhancement, in-vehicle sensing has been gaining an increasing attraction since it can continuously monitor the driver's status, from which the advanced driving assistant system (ADAS) can predict human error and thus react timely to prevent accidents from happening. In addition, it can also provide other useful real-time information about the interior of a vehicle, e.g. passenger status, or when a vehicle is parked.

Significant efforts have been devoted to in-vehicle sensing, which, according to the application scenarios, can be This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 License. For more information, see https://creativecommons.org/licenses/by-nc-nd/4.0/

classified into five categories, i.e., occupancy detection, fatigue/drowsiness detection, distraction detection, driver authentication and vital sign monitoring.

Occupancy detection [7]–[13] mainly aims to detect, localize, classify the seat occupancy states and then remind the driver before he/she leaves the vehicle. A particular case is the child presence detection (CPD) to prevent a child from being left alone in a closed vehicle, which may cause fatal damage or even death due to heatstroke [14], [15]. Existing studies about occupancy detection can be categorized into four groups according to the adopted techniques, including sensor-based [7]–[13], WiFi-based [16]–[18], imagebased [19]–[21], and radar-based [22]–[28] methods.

Fatigue/drowsiness can lead to slow reactions of a driver to the surrounding changes and has caused more than 20% of the reported accidents [29]–[31]. By enabling fatigue/drowsiness detection in the ADAS system, fewer traffic accidents can be expected, and safety and transportation efficiency can be improved. Based on the features extracted, research about fatigue detection can be roughly grouped into three types, i.e., 1) using biological signals such as electrocardiography (ECG) [32]–[48], electroencephalography (EEG) [49]–[66], electromyography (EMG) [67]–[76], 2) using facial contexts such as movement of the face [77]–[96], eye [97]–[110], etc., and 3) joint sensing of facial expressions and body/arm/leg/head motions [111]–[128].

Compared to fatigue/drowsiness, distraction can only be roughly defined since *any activity that takes a driver's attention from the driving task can cause distraction* [129] such as talking to passengers, using mobile phones, etc. As there are so many factors that may cause driver's distraction, the existing research on distraction detection mainly focuses on analyzing the driver's behaviors/activities when operating the vehicle such as acceleration/braking, and mainly contains three different groups including joint sensing of human and vehicle status [129]–[135], human sensing only [136]–[145], and cognitive sensing [146]–[148] which monitors the emotion of the driver to decipher whether he/she is focusing on driving or not.

Driver authentication [149]–[152] can help to improve vehicle security and user experience by automatically adjusting settings of the heating, ventilation, and air conditioning (HVAC), seats, and entertainment. Most of the current research in this area focuses on determining a driver's identity by jointly considering the driving behaviors and biological signals.

Driver vital sign monitoring can assist in preventing accidents caused by unpredictable sudden health deterioration of the driver as well as other in-vehicle sensing applications such as emotion sensing. Most of the conventional vital sign monitoring systems [153]–[158] require a user to wear a lot of sensor pads such as ECG/EEG, which may distract driving and thus are not applicable for driver's vital sign monitoring. Recent advances in wireless sensing techniques [159]–[163] have made contactless vital sensing possible and thus shed light on the future of driver's vital sign monitoring. The rest of the paper is organized as follows. The abbreviations used in this paper are summarized in Table I for easy reference. Section II reviews the research about occupancy detection and Section III reviews the existing works about fatigue detection. Then, Section IV summarizes the existing methods for distraction detection followed by an overview of driver authentication and vital sign monitoring in Section V. Finally, Section VI discusses the limitations and future works while Section VII concludes this paper.

# **II. OCCUPANCY DETECTION**

In-vehicle occupancy detection, which detects how many seats of a car are occupied and what object (e.g., an adults/kid/pet/inanimate item) is located at a particular seat has been a key component to enhance driving safety by the Society of Automotive Engineers (SAE) [164]. For example, knowing which seat is occupied by a passenger can be utilized to: 1) remind the passengers who are not wearing seat belts since buckling up can help to reduce the risk of fatal injuries by 45% and moderate to critical injuries by 50% [165]; 2) trigger the emergency system such as airbags in case of accidents to save lives. More importantly, leaving children, especially those who are less than 6 years old and have little ability to exit the vehicle on his/her own, alone in an unattended vehicle can cause very serious damages to organs/brain or even deaths due to heatstroke [14], [15]. As a result, enabling child presence detection has been proposed as a standard feature on the road map of the European New Car Assessment Programme (NCAP) [159], [166] to alert caregivers or emergency services if a child is left alone. Towards this end, many efforts have been devoted to developing accurate and practical occupancy detection systems. Fig. 1 summarizes the existing research about occupancy detection, which, according to the technologies adopted, are categorized into four classes as will be detailed next.

# A. SENSOR-BASED OCCUPANCY DETECTION

As shown in Table II, sensor-based occupancy detection methods [7]-[13], usually leverage different kinds of physical sensors such as weight, heat, force, capacitance, Radio frequency identification (RFID) to capture the weight, pressure, temperature, electrical continuity, capacitance, etc. elicited by the presence of passengers and then perform further occupancy analysis. This kind of methods is usually very easy to design, manufacture and deploy with affordable cost to most of Original Equipment Manufacturer (OEM) and customers. However, there are three main drawbacks of this kind of methods. First, as the equipment/sensor positions are usually pre-designed and thus fixed, they tend to suffer from very limited coverage within/next to the seats in the car. Second, it is very challenging to find a universal threshold suitable for different cars and human beings. For example, it takes different thresholds to detect the presence of people with different weights. Otherwise, it causes high false positive rate (FPR) if the threshold is too small while high false negative rate (FNR) if the threshold is too large. Third, most of them



# **TABLE I** Mapping of Abbreviations

Abbreviation	Meaning	Abbreviation	Meaning
AAA	American Automobile Association	IR	Infrared
ADAS	Advanced driving assistant systems	ISED	Innovation, Science and Economic Development
AFLW	Annotated facial landmarks in the wild	KNN	K-nearest neighbor
AR	Autoregressive	KSS	Karolinska sleepiness scale
CAN	Controller area network	LGO	Localized gradient orientation
CFO	Channel frequency offset	LRCN	Long-term recurrent convolutional network
CFR	Channel frequency response	LSTM	Long-short-term-memory
CIR	Channel impulse response	MB-LBP	Multi block local binary patterns
COTS	Commercial off-the-shelf	MDF	Median frequency
CPD	Child presence detection	MFV	Mouth feature vector
CPAM	Coordinate pair angle method	MIMO	Multiple-input and multiple-output
CNN	Convolutional neural network	MiRA	Minimum required attention
CSI	Channel state information	MNF	Mean frequency
DAQ	Data acquisition	NCAP	New Car Assessment Programme
DNN	Deep neural network	OEM	Original equipment manufacturer
EAR	Eye aspect ratio	OBD	On-board diagnostic
ECG	Electrocardiography	PCA	Principal component analysis
EEG	Electroencephalography	PERCLOS	Percentage of eyelid closure
EMG	Electromyography	PIR	Passive infrared sensor
EM	Electromagnetic	PLL	Phase-locked loops
EFV	Eye feature vector	PPG	Photoplethysmograph
EMD	Empirical mode decomposition	RF	Radio frequency
ESM	Eye screening mechanism	RGB	Red green blue
EVB	Evaluation board	RFID	Radio frequency identification
FCC	Federal Communications Commission	RMS	Root mean square
FastICA	Fast independent component analysis	RPs	Recurrence plots
FIR	Finite impulse response	RPM	Respiration-per-minute
FIS	Fuzzy inference system	RSSI	Received signal strength indicator
FLIR	Forward-looking infrared	SAE	Society of automotive engineers
FMCW	Frequency modulated continuous wave	SFO	Sampling frequency offset
FOM	Frequency of mouth	SOC	System on chip
FOV	Field of view	STO	Symbol timing offset
FPR	False positive rate	SVDD	Support vector domain description
FNR	False negative rate	SVIRO	Synthetic dataset for vehicle interior rear seat occupancy
GSR	Galvanic skin response	SVM	Support vector machine
HF	High frequency	SVMPPM	SVM-based posterior probabilistic model
HPE	Head pose estimator	SVRs	Support vector regressors
HRV	Heart rate variability	UWB	Ultra wide band
HVAC	Heating, ventilation, and air conditioning	WM	Weighted mean
IBI	Inter-beat-interval	WPT	Wavelet packet transform
IIR	Infinite impulse response	WSD	Weighted standard deviation
LOS	Line of sight	YOLO	You only look once



FIGURE 1. Related works of in-vehicle occupancy detection. (The data is obtained by searching on Google Scholar with the combinations of key words: occupancy car, occupancy vehicle, occupancy automotive, child presence car, child occupancy, and seat occupancy. We review the top 600 related papers and patents, from which we find a total of 99 ones that directly study the topic of in-vehicle occupancy detection. Accessed Mar. 06, 2022.).

Reference	Devices & Input	Methodology	Results & Advantages	Limitations
[7], [8]	Pressure sensor on the seat	Compare the object weight with a threshold	-Easy to deploy -Fast response	-Limited coverage on the seat -Unable to distinguish human from inanimate items
[13]	Pressure sensor on the buckle	Compare the forced pressure with a threshold	-Easy to implement -Fast response	-Limited coverage -Unable to work if a passenger forgets to fasten the belt
[167]	Temperature sensor inside the car	Compare the measured temperature with a threshold	-Easy to deploy -Low cost	-Performance varies in different weather conditions -Ad-hoc temperature thresholds
[10]–[12]	Capacitance sensor embedded in the seat	Measure and analyze the variation trend/pattern of capacitance	-Easy to design -Fast response -Low cost	-Performance varies from person to person due to body differences
[9]	Radio frequency identification (RFID) tag	Detect the electrical continuity within the passenger seat	-Easy to manufacture -Fast response	-Limited coverage within the passenger seat -Lack of universal criteria
[168]–[170]	Passive infrared (PIR) sensor inside the car	Measure the interior motion information such as intensity and direction	-Easy to deploy -Low cost	-Covering line of sight (LOS) with respect to the PIR sensor -Vulnerable to surrounding temperature

TABLE II	Sensor-Based	Occupancy	Detection
----------	--------------	-----------	-----------

lack the ability to distinguish human from inanimate objects. For example, weight-based approaches cannot tell apart a box from a human as long as they are of the same weight.

# **B. WIFI-BASED OCCUPANCY DETECTION**

As more vehicles are being equipped with WiFi transceivers [171]–[173], WiFi-based occupancy detection approaches [16]–[18] are becoming popular due to their superiority in cost and coverage as shown in Table III. The principle behind WiFi-based occupancy detection is that the presence or activity of a human being inside a car can

affect the WiFi signal propagation between a transmitter and a receiver, which is embedded in the channel state information (CSI) measurements and can be extracted by a dedicated algorithm. For example, Zeng *et al.* [17] proposed an approach based on statistical electromagnetic (EM) modeling, which can achieve over 96.4% detection rate with less than 3.96% false alarm and a responsive time  $\leq$  20s based on the tests over 5 real babies. [18] presented a portable CPD solution that can work on both 2.4 and 5 GHz commercial off-the-shelf (COTS) WiFi equipment by detecting biological movements at 1 – 6mm level. While WiFi-based solutions

#### TABLE III WiFi-Based Occupancy Detection

Reference	Devices & Input	Methodology	Results & Advantages	Limitations
[17]	Commercial WiFi transceiver embedded in a closed car	Detect motion and breathing of human using smart AI algorithms	-Easy to deploy -Response quickly -Low cost	-Moderate FPR for out-car motions -May miss breathing estimations thus causing miss detection
[18]	Commercial WiFi transceiver	Detect biological movement such as chest movement	-Protect user privacy -No blind-spot -Hot spot integration	-Requires about 1 minute delay -Public test results are not available due to commercial privacy

#### TABLE IV Image-Based Occupancy Detection

Reference	Devices & Input	Methodology	Results & Advantages	Limitations
[19]–[21], [176]	Two video cameras installed in the first and second row	-Video recording -Laplacian formula for edge detection -OpenCV for motion	Easy to perform	-Call for dedicated camera -Susceptible to illumination changes -May have privacy issue -Not accurate enough
[177]	FLIR <sup>1</sup> One Pro	-Image capture	-Protect the anonymity	-Call for dedicated camera
	thermal camera under	-Data augmentation	of passenger identities	-May need training for different cars
	the rear-view mirror	-Multiple-task CNN	-Low power consumption	-Lack of generalization ability
[164]	Synthetic Dataset for	-Image split for training	-Decision process is	-Only tested on 10 vehicles
	Identification <sup>2</sup> of	-Pre-train the network <sup>3</sup>	clear to end-user by	-Not accurate enough with overall
	Occupancy	-Multiple-task learning	using statistical metrics	79.87% accuracy

<sup>1</sup>Forward-looking infrared (FLIR)

<sup>2</sup>Public dataset in [178]

<sup>3</sup>Including ResNet152V2 [179], DenseNet121 [180] and EfficientNetB0-B5-B7 [181] architectures for feature extraction

enjoy low-cost and good coverage, they may suffer from distortions due to the activities outside a car such as cars passing by, since WiFi signals can penetrate the car exterior under certain conditions. Besides, other factors [174] such as channel frequency offset (CFO), sampling frequency offset (SFO), symbol timing offset (STO), and jitters of the phase-locked loops (PLLs) [175] may reduce the CSI quality and thus degrade the robustness of WiFi-based solutions.

# C. IMAGE-BASED OCCUPANCY DETECTION

To accurately estimate how many seats are occupied and further localize and recognize the objects,<sup>1</sup> image-based approaches are extensively studied [19]-[21], [164], [176], [177] because an image can provide more visible information such as the contour/edge of an object than WiFi signals. By leveraging techniques such as edge detection [19]–[21], and learning including convolutional neural network (CNN), multi-task learning [164], which can automatically identify object-related features for recognition, great performance can be achieved. However, as shown in Table IV, capturing highquality images requires dedicated cameras and it takes efforts to construct a *good* dataset to train the network, especially for manual data labeling and annotation. For better privacy protection, thermal images [177] are captured and then fed into a CCN network based on multi-task learning technique. The work in [164] designed a CNN network which is pre-trained from the existing CNN models including ResNet152V2 [179],

<sup>1</sup>Few of the WiFi-based approaches can localize and identify an object occupying a seat as the time and space resolution of COTS WiFi is limited by the bandwidth (20MHz-80 MHz) and the number of antennas ( $\leq$  3 usually).

DenseNet121 [180], and the most recent EfficientNetB0-B5-B7 [181]. The system yields about 79.87% accuracy on the public synthetic dataset for vehicle interior rear seat occupancy (SVIRO) [178] to classify people and inanimate objects over 10 different vehicle interiors and 25,000 scenarios. As seen, the accuracy is limited because different vehicles have different background information which challenges the classifier greatly.

# D. RADAR-BASED OCCUPANCY DETECTION

Recently, the unprecedented development of radar techniques [182]–[187], especially millimeter-wave (mmWave) radar has offered new opportunities for occupancy detection, classification and localization since mmWave can provide better directionality, angular, angular, and range resolution due to its high frequency and large bandwidth. As shown in Table V, recent years have witnessed the blossom of mmWave-based occupancy detection systems [22]–[28]. For example, Vayyar [22] presents occupancy detection and classification by estimating the 4D image of the object. Texas Instrument [27] demonstrates the feasibility of occupancy detection using 77 GHz frequency modulated continuous wave (FMCW) radar to construct the range-angle heatmap of the object.

Another major superiority of the mmWave system is that it is easy to be integrated on a single chip [23], [24] or a small unit [26], offering flexibility in device locations and better portability. At present, Federal Communications Commission (FCC) has been trying to enable State-of-the-Art Radar Sensors in 60 GHz Band to increase the practicality of using mobile radar devices in the 60 GHz band to perform

Reference	Devices & Input	Methodology	Results & Advantages	Limitations
[22]	Radar-on-chip 60GHz and 79GHz	-MIMO <sup>1</sup> radar -4D mmImage -Occupant classification	-High resolution -Accurate -Single-chip	-Extra hardware needed -Work in LOS condition -Complexity is hard to tell
[23]	MIMO Radar 60 GHz, FMCW	Breathing movements detection	-Seat position -Object recognition	-Extra cost -Results for awake/in-motion child are not mentioned yet
[24]	Radar EVB <sup>2</sup> 60GHz	-Remind the driver when the door is closed -2D camera	-Occupant classification and positioning -Chipset solution	-May need multiple radar sets -Multiple-sensor fusion may require high computational costs
[26]	Radar on 60-64 GHz	Radar techniques	-Approved by FCC -First sensor supplier	-To be evaluated by the market -Method not available yet
[27]	FMCW radar 77GHz	Range-angle heatmap	-Accurate within FOV <sup>3</sup> -Location of the object	-Only works for FOV in $\pm 60^{\circ}$ -Practical validation is not presented
[28]	UWB <sup>4</sup> radar	-Spatial/time features -2D convolutions -MaskMIMO	-94.6% accuracy -Lightweight -Smooth realtime run	-8 nodes distributed in the car -Need more testing scenarios <sup>5</sup> -Tests on practical human needed

#### TABLE V Radar-Based Occupancy Detection

<sup>1</sup>Multiple-input and multiple-output (MIMO)

<sup>2</sup>Evaluation board (EVB)

<sup>3</sup>Field of view (FOV)

<sup>4</sup>Ultra-wideband (UWB)

<sup>5</sup>Only testing results in the garage and outdoor parking lots are provided. More tests such as street parking are needed since passing-by cars may cause interference on the received signal and thus degrade the performance.

innovative and life-saving functions, including gesture control, detection of unattended children in vehicles [188], which provides legislative support and incentives for mmWavebased occupancy detection approaches. Companies such as Innosent [23], Infineon [24] and NOVELIC [25] have announced their system-on-chip (SOC) solution of presence detection. In February 2021, IEE VitaSens [26] launched VitaSense [189], an interior radar sensing solution for CPD in vehicles, with grant from North America and Science and Innovation, Science and Economic Development (ISED) of Canada [190], [191]. While mmWave is very encouraging, it is yet to be integrated with the current in-car system (most on 2.4 GHz and 5 GHz) without additional hardware cost.

#### **III. DRIVER FATIGUE DETECTION**

Fatigue, which degrades perception, delays reaction, and impacts judgment of a driver on his/her surroundings, has been shown as a prime culprit for over 20% of car accidents [4]. What is worse is that drivers are more prone to feeling fatigue or drowsy nowadays since the roads are becoming more crowded due to the rapid increase of motor vehicles [29] and thus the drivers have to be more focused. It is imperative to seek effective solutions for fatigue detection and prediction so that *smart cars* can sense the status of the driver and respond accordingly, such as sounding a warning/alarm message with an audio assistant system. To meet the demand, various research and commercial solutions have been proposed as summarized in Fig. 2.

## A. FATIGUE DETECTION USING BIOLOGICAL SIGNALS

By directly measuring the variation of biological response related to the human neural system, biological signal-based (e.g., EEG [49]–[66], ECG [32]–[48], and EMG [67]–[76])

226

fatigue detection has been viewed as the golden standard, and the related works are summarized in Table VI and Table VII. In most of these approaches, users are asked to wear a number of electrode pads for data collection. Then, pre-processing techniques such as Finite Impulse Response (FIR) filter, Infinite Impulse Response (IIR) filter [34], Principal component analysis (PCA) [59], empirical mode decomposition (EMD) [192] and fast independent component analysis (FastICA) [34] are adopted, which aim at removing the noise and artifacts while retaining the signal components within a certain range of frequency. Afterwards, the cleaned signal is fed into some feature extraction module to get the fatigue-related features such as  $\alpha$  and  $\beta$  information, inter-beat-interval (IBI), spatial spectrum, temporal dependency, variation, kurtosis of the power spectrum etc. To get the fatigue information, most of the existing works tend to formulate the problem as a discrete classification problem, such as support vector machine (SVM), deep neural network (DNN), K-nearest neighbor (KNN). It is worth to note that the discrete classification model is very straightforward by feeding the data into the well-studied classification models, which can usually achieve reasonably good performance. However, the manual labeling process of fatigue can be error-prone, since the evaluation is subjective and even the most experienced biological experts may get confused in distinguishing fatigue and normal status. For this reason, decision making-based on Fuzzy Inference System (FIS) [195]–[197] have been studied in assisting driver's fatigue detection since it is hard to quantify human's neuron response even for the same activity. Another main drawback of fatigue detection using biological signals is the requirement of many wearable sensors, which may distract the driver. Less intrusive sensors are being considered before such methods can be widely accepted by the market.



#### TABLE VI Biological Signal-Based Fatigue Detection-Part I

Reference	Devices & Input	Methodology	Results & Advantages	Limitations
[52]	Smartwatch with bluetooth for EEG signal acquisition	Support vector machine-based posterior probabilistic model (SVMPPM)	-91.92% detection accuracy -83.78% early detection -Evaluate by headband as the ground truth	-Wearable devices are needed -Only evaluated on simulated driving experiments
[53]	EEG sensor	- $\alpha$ and $\beta$ information extraction -Normalized Haar WPT <sup>1</sup>	Only single-channel EEG signal is needed	-User has to have physical contact with EEG sensor -Tested on public database
[54]	EEG sensor	-Hierarchical clustering to assess the subject variability -Subject-transfer framework	-Greatly reduced the calibration for about 90% -Robust over different users	Only the data from a simulated driving task is used for validation
[55]	Neuroscan EEG system with 40 electrodes covering the head area	-Temporal dependencies extraction -Spatial features fusion -Spatial-temporal CNN	-97.37% detection accuracy -High computational efficiency and low delay	Only laboratory and simulated driving data are captured for validation
[56]	EEG sensor attached on the head area	Enrich the EEG data with the intensity of head-movements	-Low-power consumption -On-chip feature extraction	6 subjects in one hour monotonous simulated driving experiment
[57]	EEG sensor attached on the head area	-Feature weighting to learn the importance of different features -Episodic training for domain generalization	Subject adaptive with no need for calibration	Details of experiment setup is not available
[58]	EEG sensor attached on the head area	-Fused multiple entropies -Autoregressive (AR) model -Channel selection	-Leave-one-out cross-validation -98.3% detection accuracy	Data is captured from 12 healthy subjects on simulated driving experiments
[59]	EEG sensor attached on the head area	Principal component analysis (PCA)+DNN <sup>2</sup>	95% detection accuracy	6 healthy volunteers in a simulated driving experiment.
[32]	ECG sensor	-Genetic algorithm and support vector machine (SVM) -Mercer kernel	-97.01% detection accuracy -Average delay of 0.55ms	Details of experiment and data acquisition are not available
[33]	Collecting ECG signal using DIY <sup>3</sup> device	-HRV <sup>4</sup> characteristics -Power spectrum ratio $\beta/(\theta + \alpha)$ -Sample entropy	-No need for patch electrodes -Realtime implementation -High integration	Experiment was performed in simulated driving conditions
[34]	-ECG electrode placed on wrist -Camera for ground truth -Manually labeling	-FIR and IIR <sup>5</sup> filters to remove the effect of noise and artifact -FastICA <sup>6</sup> to estimate the independent components -SVM and KNN <sup>7</sup>	<ul> <li>-Two-class detection provides better accuracy than 5 states<sup>8</sup></li> <li>-≥ 93.1% classification accuracy</li> </ul>	-Simulated driving data is collected only -Intrusive data collection which may distract the driver
[35]	-Wearable ECG and PPG <sup>9</sup> sensors -Polar H7 strap and a PPG sensor	-Recurrence plots (RPs) generated from the R–R intervals (RRIs) of heartbeats as features of CNN	Overcome interference from the slight movement of subjects on data acquisition	Only data collection from a virtual driving environment
[36]	-Infrared (IR) camera -LAB DAQ <sup>10</sup> device -ECG sensor with Ag–AgCl electrodes	-Higher order spectral feature -Linear discriminant analysis -Quadratic discriminant analysis	Overall maximum accuracy of 96.75%	Driving monotonously on a driving simulator at a limited speed for long hours to simulate fatigue
[67]	EMG, GSR <sup>11</sup> and bluetooth for data acquisition	Variations of EMGs in frequency-domain	-92% detection accuracy -KSS <sup>12</sup> for ground truth labeling	No practical measurement is available
[68]	Surface EMG signal from upper arm and shoulder	-Five features including range, variance, spectral power and kurtosis, and shape factor are fed to a KNN classifier	90% detection accuracy over 13 testers	Driving simulator on a monotonous route

<sup>1</sup> Wavelet packet transform (WPT)

<sup>2</sup> Deep neural network (DNN)

<sup>3</sup> Do it yourself (DIY)
 <sup>4</sup> Heart rate variability (HRV)

<sup>5</sup> Finite Impulse Response (FIR), Infnite Impulse Response (IIR)

<sup>6</sup> Fast independent component analysis (FastICA)

<sup>7</sup> K-nearest neighborhood (KNN)

<sup>8</sup> Normal, drowsy, fatigue, visual and cognitive inattention

<sup>9</sup> Photoplethysmograph (PPG)

<sup>10</sup> Data acquisition (DAQ)

<sup>11</sup> Galvanic skin response (GSR)

<sup>12</sup> Karolinska sleepiness scale (KSS)



(d) Overview of the existing work.

FIGURE 2. Related works of in-vehicle fatigue detection. (The data is obtained by searching on Google Scholar with the combinations of key words: driver fatigue, driver drowsiness, vehicle fatigue, vehicle drowsiness, automotive fatigue, and automotive drowsiness. We review the top 600 related papers and patents, from which we find a total of 189 ones that directly study the topic of in-vehicle fatigue detection. Accessed Mar. 11, 2022.).

#### TABLE VII Biological Signal-Based Fatigue Detection-Part II

Reference	Devices	Methodology	Results & Advantages	Limitations
[69]	DIY EMG data collection system with electrode on the surface of cloth with thickness $\leq 2$ mm	-Special design of amplifier -FastICA and digital filter -Kolmogorov Smirnov Z analysis -Mahalanobis distance	Noncontact data acquisition from biceps femoris	Driving simulator on a monotonous route
[70]	EMG sensor collecting data during driving a truck	Mean frequency (MNF), median frequency (MDF), and the signal RMS <sup>1</sup> amplitude	10 healthy volunteers	Physical contact with users
[192]	Non-contact EMG sensors located in a cushion on the driver's seat	-FastICA, denoise by EMD <sup>2</sup> -PCA and multiple linear regression study linear relation among features	-12 healthy males -Model based -91% detection accuracy from 13 testers	Simulated driving experiments
[193]	-Wearable chest-strip with sensor -HRV estimation	The weighted mean (WM) and the weighted standard deviation (WSD) of the high frequency (HF) band in the power spectrum	Results of SVM, random forest and KNN are demonstrated	-Virtual driving environment -Physical contact with user for data collection which may distract the driver
[51]	ECG, EEG and PSG <sup>3</sup> sensor are used for HRV and ground truth data collection	-Eight HRV features are monitored -Multivariate statistical process control	-92.3% detection accuracy over 34 testers -1.7 times false alarm per hour	-Data collected in a virtual vehicle simulator -Physical contact needed for data collection
[194]	-Cheststrap -Wrist watch	Multi-layer artificial neural network and SVM	91.3% detection accuracy	Wearable sensors

<sup>1</sup>Root mean square (RMS)

<sup>2</sup>Empirical mode decomposition (EMD)

<sup>3</sup>Polysomnography (PSG)



#### **TABLE VIII** Fatigue Detection Using Independent Facial Features

Reference	Devices & Input	Methodology	Results & Advantages	Limitations
[77]	-Camera for self-built data set -Public data set WIDER FACE	-Improved YOLOV3-CNN for facial region detection -Eye feature vector(EFV) and mouth feature vector(MFV)	-Detect the fatigue state at a speed of over 20fps with 95.10% accuracy -50 test drivers	Simulated driving experiments
[78]	-National Tsing Hua University Driver Drowsiness Detection (NTHUDDD) dataset -Video clips for labeling	-Multiple-task cascaded CNN -Static features extraction from a partial facial image -Dynamic features extraction from a partial facial optical flow -Two-stream neural network	An accuracy of 97.06% over 36 subjects of different ethnicities on a public dataset	-Practical validation is not provided yet -Privacy concern, which is common for all image based methods
[79]	-57 normal videos, 94 slightly fatigued videos and 49 severely fatigued videos.	-Face key point detection -Multi block local binary patterns (MB-LBP) -Adaboost classifier -Fuzzy inference system	$-\geq 94.7\%$ accuracy -Detection speed is 53 frames/s	When the driver wears glasses or the face rotates at a large angle, the accuracy of the algorithm decreases
[80]	YawDD dataset [88] and self-built dataset with 10 volunteers	-Multiple-task CNN for face detection -DLIB <sup>1</sup> to localize the key points of face -LSTM <sup>2</sup> for classification	-≥ 93% accuracy -Reduce half of the running time than DLIB	-The detection performance under insufficient light still degrades -System vulnerable to light conditions
[81]	-Extended Cohn-Kanade dataset -Psychological image at stirling dataset	-Facial landmarks -OpenCV for image processing -DLIB for feature extraction	82.79% detection accuracy	No practical validation is provided
[97]	-CEW Database -ZJU Database -BUAA Eye Database	-Dual-stream bidirectional CNN -Eye gaze pattern analysis -Eye screening mechanism (ESM) to eliminate the detected errors	Improves about 2.9% in the average accuracy compared with using CNN alone	No practical validation is provided
[98]	-Infrared videos for self-built data base -ZJU Database	-CNN for eye state detection -PERCLOS and blink frequency	Robust for wearing glasses	Need active infrared light (850nm) to fill light illumination
[99]	Raspberry Pi 4	Eye aspect ratio (EAR) technique	-90% detection accuracy -Under initial, wearing spectacles, dim light and microsleep condition	Not tested over enough number of subjects
[100]	Head-mounted eye-tracking camera	Fuzzy KNN	Reach to about 89% accuracy in average	Highway-driving simulator experiment
[101]	-Dome camera installed on the bus -23 testers under different lighting conditions	-Eye openness detection using spectral regression -Adaptive integration to estimate eye state	Explore the case when a camera is in an oblique view w.r.t. the driver's face	Validation is on simulated bus driving videos

<sup>1</sup>DLIB:an open-source software library

<sup>2</sup>Long-short-term-memory (LSTM)

# **B. FATIGUE DETECTION USING INDEPENDENT FACIAL** FEATURES

Without requiring wearable sensors, image/video-based fatigue detection using facial features has become popular, such as those based on face recognition [77]–[96], eye detection [97]–[110], and the combination of the features extracted from face, eye, mouth, etc. In most of these approaches, a face/eye/mouth region detection module is firstly designed to refine the input image to remove redundant information outside the region of interest. As shown in Table VIII, region recognition methods include you only look once (YOLOv-CNN) [198], multi-task cascaded CNN, DLIB keypoint detection [199], etc. The next step following the region detection is to extract the fatigue-related visible features such as eye open/close/gaze, mouth open/close, face being twisted or not. Afterwards, the joint analysis of the extracted features is performed. For example, the percentage of eyelid closure over the pupil over time (PERCLOS) of a driver larger than 80% [200] is a strong indication that he/she is drowsy, even though the specific threshold/percentage may vary from person to person and at different time over a day. In this case, by continuous monitoring, if the system further detects that the driver yawns more frequently than usual, there is a high probability that he/she is sleepy and thus an alert can be triggered. In the last, different strategies can be adopted such as Two-stream neural network, Adaboost classifier, Fuzzy inference fusion, and long-short-term-memory (LSTM) network to output the final decision.

While many related works have been proposed with their own advantages and drawbacks, as shown in Table VIII,

they share several common limitations: 1) Putting a camera in front of the driver during driving may not only induce privacy concern but also distract the driver and thus increase the risk of accidents; 2) Many of the related works are studied on public datasets such as WIDER FACE [77], National Tsing Hua University Driver Drowsiness Detection (NTHUDDD) dataset [78], [112], [201], YawDD dataset [88], CEW Database [97], ZJU Database [97], BUAA Eye Database [97], most of which are collected in a laboratory environment when the driver is driving on a simulator. As a result, it is really hard to conclude about the practical performance since human beings tend to have very different biological reactions in practical driving. To expedite the real-world application, efforts are still needed on developing highly efficient data collection tools on practical driving for further validation; 3) Most of the existing studies are based on the dataset when users face right towards the camera, while few of them have discussed the case if a user faces towards the camera with an oblique angle, since driving can involve frequent activities requiring head turns, such as checking the rear-mirror, looking at the side mirrors before lane changes, etc.

# C. FATIGUE DETECTION USING HYBRID ANALYSIS

As aforementioned, to capture high-quality images/videos so that minute facial/eye/mouth changes can be extracted, dedicated cameras are needed. In addition, under some circumstances, a strict installation angle/position is required to make sure that the camera and user's face are facing towards each other. However, a driver has to keep checking the surrounding environment during practical driving, the relative position/angle between the camera and the driver's face is time variant, and it is impractical to assume that frontal images are always available. To tackle this issue, some of the existing works have explored solutions by studying how the big motion of a driver, such as head movement due to nod, arm/hand motion when moving their hands away from the steering wheel unintentionally, can assist the fatigue detection, because sensing big motions is always feasible during driving. Although relying on sensing the big motion itself may not be accurate enough for reliable fatigue detection as such big motions can also happen when the driver is sober, they can provide good auxiliary information.

Towards this end, many research works [111]–[128] have been proposed by jointly analyzing the big motions and the imperceptible changes corresponding to the subtle motion of face/eye/mouth as shown in Table IX. For example, Part *et al.* [111] presented a joint analysis on local facial expression and head gesture using VGG-FaceNet and FlowImageNet architecture, respectively. The results show that joint features including both face and head can contribute about the 5% improvement in drowsiness detection accuracy over the public NTHUDDD video dataset [112], which are collected over 36 subjects including different genders and ethnicities. While [111] is evaluated on a public dataset collected under simulated driving conditions, Mittal *et al.* [113] develops a fatigue detection system that combines the information from head pose (with a particle-filter based 3D model to track head motion), lip and eyes, which is then comprehensively evaluated with 14 subjects driving a car on different round-trip routes through the University of California campus at different times, including morning, afternoon, dusk, and night. Further studies also involve the head pose dynamics [114], [115], [202], head orientation and arm position [116], [117], which improve the fatigue detection accuracy by 2% - 10%, compared to the benchmark methods using only facial features. To further handle the time-variant relative position between the camera and a driver's face, multiple cameras can be distributed around the car for data acquisition [114], [115], [202]. Although the fatigue detection accuracy is improved, many new practical problems arise as well, such as the cost of hardware, deployment, computational complexity, and more importantly how to fuse the information from multiple cameras while satisfying the real-time detection requirement.

#### **IV. DRIVER DISTRACTION DETECTION**

Driver distraction, which can increase the risk of accidents, may be caused by many factors. As introduced in Section I, there exists no universal definition for distraction during driving [205]-[207], and a widely accepted concept is that, any activity that takes a driver's attention from driving belongs to the cause of distraction [129] such as talking to passengers, using mobile phones, under different kinds of negative emotions including anger [208], anxiety [209], sadness [210], etc. As shown in Fig. 3, the community has been exploring how to detect/prohibit distraction in many different directions to improve driving safety. One example is that playing with mobile devices when driving is legislatively prohibited in most countries. Besides, car manufacturers are adding more convenient designs such as integrating switches of phone-call, music-playing, cruise-setting on the steering wheel area so that a driver does not need to move his/her hand off the steering wheel when they have to utilize related functions. Also, some of the amusement features are disabled during driving such as that Tesla [211] stops allowing drivers to play video games during driving. However, there exists a conflict between simplifying the design/functions and satisfying the users. In other words, to make the car more intelligent and improve the driving experience, manufacturers have to develop and integrate more functions (e.g., entertaining, relaxing), which again will increase the chance of distraction. Therefore, an automatic distraction detection system is needed, which can alert a driver, or more intelligently, provide real-time corrections whenever distraction is detected.

# A. DISTRACTION DETECTION BASED ON JOINT SENSING OF HUMAN AND VEHICLE

Among the many distraction detection studies [129]–[135], joint sensing of the vehicle and human status is firstly proposed, as shown in Fig. 3 and Table X. Tang *et al.* first [130] presented a driver's distraction system by leveraging the vehicle data (usually including speed, steering angle, position of



#### TABLE IX Fatigue Detection Using Hybrid Analysis

Reference	Devices & Input	Methodology	Results & Advantages	Limitations
[4]	-YawDD video dataset [204] -NTHUDDD video dataset [203]	-Eye and mouth characteristics -Multiple-task learning -PERCLOS and FOM <sup>1</sup>	98.81% fatigue detection accuracy	-Privacy issue -More practical validation needed
[111]	NTHUDDD video dataset [203]	-Face and head gesture based on FlowImageNet -Facial feature representation based on VGG-FaceNet	-Performance comparison with CNNs and LRCN <sup>2</sup> -Evaluations on NTHUDDD dataset with 36 subjects	No validation on real driving conditions
[113]	-IEEE1394 camera to capture the gray-scale video -Vicon optical motion capture to track the head pose	-AdaBoost, LGO <sup>3</sup> histogram -Support vector regressors (SVRs) -Particle filter to track the 3-D head motion	-Head pose, lip corner, eye -14 subjects driver the LISA-P <sup>4</sup> for experiments -Different round-trip routes at different times	-Near-IR illuminator to ensure the light conditions -Performance degrade when the yaw of head approaches 90°
[114], [115], [202]	Distributed camera framework	-Head pose and dynamics -Facial features + geometric relationships to estimate the head pose using a 3-D model	-Practical on-road driving in urban streets and freeways -Emphasis on events inducing spatially large head movements (e.g., merge and lane change)	-Call for multiple cameras inside the car -Fusion among multiple cameras may increase the complexity in practical applications
[116]	Eye tracking and accelerometer to capture head motion	-Blink rate, yawn motion, head motion -AdaBoost and multi-nomial ridge regression	-Computer game driving -96% and 90% accuracy for within and across subjects -Revealing a counterintuitive observation that drivers yawn less before fall asleep	No practical driving validations
[117]	Kinect active sensor	-Eye behavior + arm position + head orientation + facial expressions -AdaBoost classifier and Hidden Markov model	-Driving simulator over 8 drivers -90% distraction detection accuracy	-Manually labeling is time consuming and error-prone -Practical tests needed

<sup>1</sup>Frequency of mouth (FOM)

<sup>2</sup>Long-term recurrent convolutional network (LRCN)

<sup>3</sup>Localized gradient orientation (LGO)

<sup>4</sup>Laboratory for intelligent and safe automobile (LISA)

the accelerator pedal, the brake pedal, etc.) gathered from the vehicle controller area network (CAN) [204] system, which is then fed into an SVM classifier for distraction detection. Later, motion information which mainly corresponds to the big motion of the human body such as body/leg/arm movements were further involved in [131] and yielded about 90% detection accuracy. To further improve the accuracy, the relationship between the head motion and distraction was studied in [132] and then fused with the vehicle data [134], which also explores the time-domain information by utilizing LSTMrecurrent neural network. The correlation between eye glance and steering movements was analyzed in [212], which verified the feasibility of distinguishing different types of distraction. To test the performance, a real-time system was implemented in [129] and 30 participants were recruited to drive on a straight country road while performing eight pre-defined secondary task (e.g., playing radio, setting the navigation to a destination) on the multimedia interface to evoke distractions. In total, the authors got about 150 minutes distraction data and 50 minutes attentive data, which demonstrates about 96.6% accuracy.

While these works have shown promising results, most of them are evaluated on the data collected from a driving simulator or practical driving but following a simple route. Although many efforts have been made to make the driving simulator more realistic such as involving challenging routes, playing sound around as distractions, experiences from a driving simulator are still different from practical driving [147], and the validations/findings from the aforementioned works may not hold in practice. as shown by the degraded performance during practical driving [129], [134], and it is worthwhile to conduct more real-world data based studies.

#### **B. DISTRACTION DETECTION ON HUMAN SENSING**

Instead of joint vehicle and human sensing, distraction can also be detected based on human sensing only. The main reasons are as follows. First, vehicle data is not always available. Second, it is difficult to build a universal vehicle data-based driving profile since driving behavior can be affected by many external factors and a driver may respond differently to the same stimulus. For example, braking frequency on a highway



(d) Distribition of different categories in different years.

**FIGURE 3.** Related works of in-vehicle distraction detection. (The data is obtained by searching on Google Scholar with the combinations of key words: driver distraction, driver inattention, driver behavior, vehicle distraction, vehicle inattention, automotive distraction, and automotive inattention. We review the top 700 related papers and patents, from which we find a total of 168 ones that directly study the topic of in-vehicle distraction detection. Accessed Mar. 15, 2022.).

#### TABLE X Distraction Detection by Joint Sensing of Vehicle and Human

Reference	Devices & Input	Methodology	Results & Advantages	Limitations
[130]	-Prototype vehicle CAN <sup>1</sup> bus -Dynamic vehicle data	-Non-intrusive and real-time detection -Driving with secondary tasks	Up to 95% detection accuracy	Data is collected by using a static driving simulator
[131]	-Vehicle CAN data -Motion sensor	Focus on perceivable distractions with leg and head movements	-Over 90% detection accuracy -Real-time testing	Testing route is limited
[132]	FaceLAB eye tracker [133]	-Eye and head movements -Laplacian SVM and semi-supervised extreme learning machine	Semi-supervised learning improves about 2.45% using unlabeled data	-Constrained experiment settings with limited driving behaviors -Performance on unsupervised learning needs to further evaluate
[129]	-CAN-Bus data -Head tracking images -150 minutes distracted and 50 minutes attentive data [206]	-Joint use of head and the information of a car itself -Long-range temporal context of driving and head tracking data -LSTM	-Accuracy of up to 96.6% -30 participants (12 female and 18 male, aging from 23 to 59)	-Constrained distracted activities during driving -Performance degrades under complicated driving behaviors -Straight route with moderate traffic density
[134]	-Stereo vision system -Lane position sensor -CAN vehicle sensor	-SVM -Attention mapping algorithm	-12 professional drivers driving Volvo FH 12 truck -Accuracy varies from 10% to 99%	Performance varies a lot due to different relative angles between the camera and driver

<sup>1</sup>Controller area network (CAN) [204]



Reference	Devices & Input	Methodology	Results & Advantages	Limitations
[213]	-Full scale image -AUC <sup>1</sup> Distracted Driver dataset	Bidirectional-LSTM (BiLSTM) Networks	Classification accuracy of 92.7%	Practical data validation is not provided
[214]– [216]	Track and record the duration of looking on or away from the road	Context dependency and the possibility to self-pace from MiRA <sup>2</sup>	A real-time implementation is demonstrated	Driving simulator experiment with 16 bus drivers
[217]	Public database [221] with full scale image inside the car	-Supervised learning -Combination of CNN and random forest	Detection accuracy of 95%	Public database under controlled experiment
[219]	Driving simulator with data acquisition using Matlab simulink platform	-Euclidean distance -Fuzzy logic and fuzzy neural networks	-Driver-in-the-loop experiments -Within 20% estimation error	Practical driving test is not provided
[220]	Self-built dataset based on AUC dataset model	-Encode and decoder -Capsule network -Face and hand detection	90% accuracy with 92% precision score	The routine condition during the experiment is not presented

#### TABLE XI Distraction Detection by Sensing of Human Only

<sup>1</sup>American University in Cairo (AUC) Distracted Driver Dataset

<sup>2</sup>Minimum Required Attention (MiRA)

and urban roads is different, while driving in snowy weather is different from driving on a sunny day as well. In addition, the measurement error of the CAN system is vehicle-dependent, which again induces new noise in the vehicle-specific dataset.

Distraction detection based on human sensing only can be mainly divided into two categories. Existing works in the first category mainly leverage the visualized big motion (e.g., head pose/orientation [136]-[139]) of the human body or the minor motion involved in the eye movement/glance [140]-[142], facial expression [143]–[145], and etc. For example, Zhao et al. [136]-[138] introduced a distraction detection method by utilizing the head pose estimator (HPE \_ Resnet50) network structure to extract the head pose/orientation (described in Euler angle) of a driver [136], based on the 300W-LP [221] and Annotated facial landmarks in the wild (AFLW) datasets [222]. A similar idea was proposed in [137], [138], which extracted the head pose using a coordinate-pair-anglemethod (CPAM) and then DNN for further classification. Praveen et al. [223] demonstrated the feasibility of distraction detection by tracking the face pose using a clustered approach based on Gabor features. The full-scale information of a human body was leveraged by an ensemble of ResNets in [224] to distinguish distraction from images of normal driving, yielding an accuracy of 94.28% on the American University in Cairo (AUC) dataset. On the other hand, Rezaei et al. [142] created a cascaded network using Haar-like Masks to detect the subtle eye movement such as opening/closing for distraction recognition, which could detect distraction from both the frontal direction and an oblique angle of a tilted head pose, making a big step towards practical applications. [141], [216], [225] explored metrics such as eye gaze direction [225], blink pattern [141], and on-road/off-road gaze duration [216]. [226]-[228] demonstrated the feasibility of using activities of eyes and mouth, and a review of driver distraction detection using facial expressions was presented in [144], [145]. Note that the different features extracted from

e, Zhao follows: *Step 1:* Image capturing using dedicated RGB/thermal cameras, which are usually mounted on the wind-shield/dashboard and pointed towards a driver's face as

much as possible while not blocking his/her view. *Step 2:* Image pre-processing, such as resizing, noise removal, enhancement, and region detection corresponding to the driver's arms, legs, hands, head, torso, face, mouth, eyes, etc. Note that different region detection methods may be designed for a specific purpose. For example, eye detection takes a smaller window to capture more details while body contour extraction requires full-scale images.

subtle eye/face/mouth movements are usually fused [143] and

then fed into various classifiers such as AdaBoost, Random

Instead of using either the big motion [136]–[139] or minor

motion [140]–[145], [226]–[228], the second category [140],

[146], [203], [213]–[220], [229]–[231] fuses features from the

big motion and minor motion together as shown in Fig. 3

and Table XI. Although different features are extracted, the main steps of this kind of methods can be summarized as

Forest, SVM, CNN, DNN for distraction analysis.

• *Step 3:* Feature extraction, construction, and classification, which are closely related to the cascaded network and the loss/objective function adopted. The two most common feature extraction methods are shape-based (e.g., calculating the *distance* such as Euclid distance using several key points in the image) and appearance-based by leveraging the color, context, or correlations between different images.

# C. COGNITIVE DISTRACTION DETECTION

Another type of distraction is cognitive distraction, which is mainly caused by negative emotions of a driver such as anger [208], sadness [210], and anxiety [209]. Since emotion is mainly related to the activity of one's brain and neural system, cognitive distraction is hard to be detected using the aforementioned detection approaches. It is possible that one may have different behaviors under different moods. However, different people have different ways of expressing their emotions, and thus it may not be accurate to judge one's emotion purely based on his/her behavior. In this sense, cognitive distraction is probably the most difficult type of distraction to be detected [146].

Recent work [147] presented a review of the existing research on in-vehicle emotion sensing, which, according to the information adopted to sense emotion, can be divided into biological signal based (e.g., ECG, Heart rate and blood pressure, etc.) [232]-[239], speech signal based [240], [241], facial expression based [242]-[248], behavior based [249], and those using the combination of different features [148]. Biological signals-based methods can achieve good accuracy because those signals are directly related to the physiological response of a human being under different emotions. However, capturing biological signals is usually intrusive and requires physical contact between electrodes and human body, and is not convenient for a practical driving scenario. In addition, biological signals are often very weak and thus highly vulnerable to external distortions such as noise and unavoidable human body motions. Note that Du et al. [148] have shown that the joint use of biological features (e.g., heart rate extracted from RGB images) and facial expressions from images can improve the emotion detection accuracy by about 5%, which may shed light on a new direction of cognitive distraction detection.

#### V. MORE APPLICATIONS OF IN-VEHICLE SENSING

In this section, we introduce two more other in-vehicle sensing applications, especially those wireless sensing-based techniques due to its superiority in cost and coverage.

# A. DRIVER AUTHENTICATION

As remote keyless system [250] has become standard equipment for modern vehicles, most of them are still relying on a token matching and rolling scheme, which has been reported for several security concerns [150], [251]–[253]. Therefore, enabling a smart driver authentication system can help to protect a car from improper use without the permission of the driver/owner [150]. Moreover, automatic driver authentication can enable intelligent driver-specific adjustments, such as the seat and mirror positions [149].

Towards this end, the works [150], [151] built a driver identification system by sensing the driving behavior using the data from the in-vehicle CAN system, with SVM and CNN + SVDD (Support Vector Domain Description) classifiers, respectively. Face recognition techniques are leveraged in [152] while [254] utilizes the On-Board Diagnostic (OBD) port for collecting data about speed, pedal movement, fuel flow, etc., which are then fed into a machine learning module for classification. Biometric-based driver authentication methods have also been proposed using different biometric information including palm prints and veins [255], brain waves [256], and combinations of hand swipes, voice, and faces [257]. The authors in [149] presented a driver identification system by recognizing the unique radio biometric information [258] embedded in the CSI of commercial WiFi. A long-term driver radio biometric database was built to train a generalized DNN that is robust to the environment changes, and experiments demonstrate up to 99.13% accuracy.

# **B. DRIVER VITAL SIGN MONITORING**

In-vehicle health/vital sign monitoring has also become attractive recently, because vital sign signals such as heart rates can help to improve other in-vehicle sensing functions such as emotion sensing [148]. Also, continuous health monitoring can reduce the risk of accidents in unpredictable and imperceivable health deterioration (such as a sudden pathological attack or heart stroke, which is difficult to be detected based on emotion or behavior sensing) of a driver when he/she is driving.

Existing works on driver vital sign monitoring include the sensor-based methods [153]-[158], vision-based methods [259]-[262] and radio frequency (RF)-based methods [263]-[275]. The sensor-based methods require wearable sensors such as photoplethysmography (PPG) [156], ECG [154], [155], EEG [153], [276], voltage-controlled oscillators [157], and electromagnetic coupled sensor [158] to capture physiological signals for vital sign analysis. They are accurate due to the direct contact with a human body but tend to be cumbersome, uncomfortable, and distracting for a driver when driving, thus hindering practical applications. Vision-based methods [259]-[262] which usually leverage a camera mounted inside a car to capture images/videos for vital sign analysis, are less intrusive by reducing physical contact than sensor-based methods, but raise privacy concerns and are susceptible to illumination conditions, which again inhabits the wide deployment. More recently, RF based vital sign sensing systems have been gaining more interests since they do not require any wearable sensor while preserving user privacy and robustness over different illumination conditions. Intuitively, RF signals reflected off human subjects will be modulated [277]-[282] by body movements including chest and heart movement due to respiration and heartbeat. As a result, one can decipher the vital sign information embedded in the received RF signals without any intrusion to a driver.

Currently, WiFi- [159], [160] and mmWave-based [283], [284] systems are the two mostly adopted RF-based approaches for in-vehicle vital sign monitoring. For example, Wang *et al.* [160] presented a WiFi-based multi-person (up to 4, a typical number of total passengers in a car) respiration rate estimation system with subcarrier selection and trace concatenation, which yields up to 98.9% detection accuracy with the respiration rate estimation error less than 1 respiration-per-minute (RPM). Moreover, it [160] also explored the feasibility of people recognition using the distribution of the respiration estimations for a certain period. Although WiFi-based vital sign sensing methods [160], [280], [281] have shown great advantages in coverage, low-cost, and excellent portability by reusing the existing on-car WiFi, they



lack good spatial resolution for reliable heart rate sensing. As an alternative, mmWave [163], [264]–[268], [270]–[275] has shown superior spatial resolution because it operates in a much higher frequency, with larger bandwidth and higher integration capability to equip more antennas on a single chip, and many mmWave-based vital sign monitoring systems have been proposed. For example, the works [163], [264]–[266] (although not for in-vehicle sensing) have demonstrated the feasibility of using mmWave to extract breathing and heart rates simultaneously [163], [265] and further estimation of heartbeat variability is presented in [267], [268], [283].

Note that the aforementioned mmWave-based vital sign sensing may not be directly applied to a driver's vital sign monitoring when he/she is driving because the vital signs are very weak and thus easy to be overwhelmed by motions involved in driving. Recently, by exploring the 2D-correlation of the range-angle heat map of the received RF signal, Wang *et. al.* [271] proposed a motion compensation method to mitigate the impact of interfering motions on driver vital sign monitoring when driving by aligning and then concatenating the vital signals in different time intervals dynamically. Extensive experiments show an estimation accuracy of 99.17%, 98.94% and 94.11% for respiration rate, heart rate, and inter-beat-interval estimations, respectively.

### **VI. DISCUSSION**

Despite the significant achievements for in-vehicle sensing applications, a number of issues still remain open for future studies. In this section, we share several possible research opportunities for interested readers.

# A. EVALUATION OF THE SYSTEM

According to the surveyed approaches, there are two common limitations in evaluation. First, most of the published works are evaluated on the data collected either under simulated driving environments or practical experiments along with simple routines. Although many efforts have been made to make the simulated experiments as natural as possible, knowing a mistake does not really hurt, human beings under simulated driving experiments will have much different physiological responses from that of real-world driving [147], [285]. As a result, it is questionable whether the existing research can be generalized to practical driving. Second, different methods are usually evaluated on different datasets, and it is difficult to judge which one is better by just comparing the related approaches side-by-side because even a small difference in data may affect the performance especially for data-driven approaches. Therefore, it is worthwhile to develop highly-efficient in-vehicle sensing data collection platform and build more standard public datasets for comparison across different methods.

## **B. FUSION OF DIFFERENT FEATURES**

Many of the existing studies [129], [131], [134], [135] have shown that joint sensing over different features together can improve performance. However, few of them have studied how much extra cost it takes during the fusion process. For VOLUME 3, 2022 example, to train a network that can leverage sensing features from both big motions (e.g., head/leg/arm) and small motions (e.g., eye open/close) may take twice or even higher computation and memory than that of utilizing just one of the features. This is because that the network may suffer from the *Curse* of Dimensionality [286] with the increment of the number of features. Hence, the efforts needed to construct the dataset and then train the network may grow exponentially. Therefore, optimization of feature fusion is important for in-vehicle sensing.

## C. PERSONALIZED IN-VEHICLE SENSING

Most of the current in-vehicle sensing studies aim at improving the safety. However, with the development of automotive techniques, drivers may expect to be able to adjust the sensing functionality freely. For example, an elderly driver may want the sensing system to pay more attention to his/her own health status during driving, while another driver who has a young baby on board cares more about his/her baby on the back seat. Thus, personalized in-vehicle sensing which can meet the various requirements on different functions may be of interests.

#### **VII. CONCLUSION**

This paper presents a survey on the state-of-the-art in-vehicle sensing technology. We classify the existing research works into five topics, i.e., occupancy detection, fatigue/drowsiness detection, distraction detection, driver authentication, and vital sign monitoring. We discuss the motivation and main techniques adopted in each topic, explain how these techniques are leveraged, and analyze the limitations and possible future solutions. A high-level discussion about the evaluation and feature fusion is provided to narrow the gap between theoretical research and practical applications. Personalizing in-vehicle sensing is also covered which may inspire more research to improve driving safety while making driving experience more customized.

#### REFERENCES

- "Automotive revolution perspective towards 2030," Accessed : Mar. 06, 2022. [Online]. Available: https://www.mckinsey.com/industries/ automotive-and-assembly/our-insights/disruptive-trends-that-willtransform-the-auto-industry/de-DE
- [2] "Automotive and mobility statistics," Accessed : Mar. 06, 2022. [Online]. Available: https://movotiv.com/statistics#:\: text=on%20Global%20Journeys%20%26%20Markets%20(pre% 2DCovid)&text=The%20average%20trip%20duration%20globally, a%20vehicle%20over%20their%20lifetime
- "Americans spend 70 billion hours behind the wheel," Accessed : Mar. 06, 2022. [Online]. Available: https://newsroom.aaa.com/2019/ 02/think-youre-in-your-car-more-youre-right-americans-spend-70billion-hours-behind-the-wheel/
- [4] B. K. Savaş and Y. Becerikli, "Real time driver fatigue detection system based on multi-task connn," *IEEE Access*, vol. 8, pp. 12491–12498, 2020.
- "Road traffic injuries," Accessed : Mar. 06, 2022. [Online]. Available: https://www.who.int/news-room/fact-sheets/detail/road-trafficinjuries
- [6] "What percentage of auto accidents are caused by human error?," Accessed : Mar. 06, 2022. [Online]. Available: https://crashstats.nhtsa. dot.gov/Api/Public/ViewPublication/812456

- [7] A. R. Marc, "Warning system for detecting presence of a child in an infant seat," U.S. Patent 5,949,340, Sep. 7, 1999.
- [8] J. C. Charles, "System to detect the presence of an unattended child in a vehicle," U.S. Patent, 7,170,401, Jan. 30 2007.
- [9] J. K. Lee, D. R. Spach, J. K. Srinivasan, and D. p. Umstead, "Wireless system to detect presence of child in a baby car seat," U.S. Patent, 7,321,306, Jan. 22 2008.
- [10] J. Albesa and M. Gasulla, "Occupancy and belt detection in removable vehicle seats via inductive power transmission," *IEEE Trans. Veh. Technol.*, vol. 64, no. 8, pp. 3392–3401, Aug. 2015.
- [11] A. Ranjan and B. George, "A child-left-behind warning system based on capacitive sensing principle," in *Proc. IEEE Int. Instrum. Meas. Technol. Conf.*, 2013, pp. 702–706.
- [12] B. George, H. Zangl, T. Bretterklieber, and G. Brasseur, "Seat occupancy detection based on capacitive sensing," *IEEE Trans. Instrum. Meas.*, vol. 58, no. 5, pp. 1487–1494, May 2009.
- [13] L. Davis, "Child carseat alert system," U.S. Patent 7,250,869, Jul. 31 2007.
- [14] M. Catherine, N. Jan, and Q. James, "Heat stress from enclosed vehicles: Moderate ambient temperatures cause significant temperature rise in enclosed vehicles," *Pediatrics*, vol. 116, no. 1, pp. e109–e112, 2005.
- [15] J. Null, "Heatstroke deaths of children in vehicles," Accessed : Mar. 06, 2021. [Online]. Available: https://www.noheatstroke.org/original/
- [16] Q. Xu, B. Wang, F. Zhang, D. S. Regani, F. Wang, and K. J. R. Liu, "Wireless AI in smart car: How smart a car can be?," *IEEE Access*, vol. 8, pp. 55091–55112, 2020.
- [17] X. Zeng, B. Wang, W. Chenshu, S. D. Regani, and K. J. R. Liu, "Intelligent Wi-Fi based child presence detection system," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process.*, 2022, pp. 11–15.
- [18] UniMax Electronics Inc., "Unimax CPD solution," Accessed : Mar. 06, 2022. [Online]. Available: https://www.unimax.com.tw/en/ product\_detail/76
- [19] I. H. Muhamad and F. R. Rashidi, "In-car suffocating prevention using image motion detection," *Recent Adv. Elect. Eng. Ser.*, no. 12, pp. 145–150, 2013.
- [20] H. Cai, D. Lee, H. Joonkoo, Y. Fang, S. Li, and H. Liu, "Embedded vision based automotive interior intrusion detection system," in *Proc. IEEE Int. Conf. Syst., Man, Cybern.*, 2017, pp. 2909–2914.
- [21] R. Panda and A. K. Roy-Chowdhury, "Multi-view surveillance video summarization via joint embedding and sparse optimization," *IEEE Trans. Multimedia*, vol. 19, no. 9, pp. 2010–2021, Sep. 2017.
- [22] Vayyar Inc., "Testing with all ages of infants in various child seat," Accessed : Mar. 06, 2022. [Online]. Available: https://vayyar.com/auto/
- [23] InnoSenT Inc., "Child presence detection with RADAR anonymous vehicle interior monitoring," Accessed : Mar. 06, 2022. [Online]. Available: https://www.innosent.de/en/automotive/incabinradar-monitoring/
- [24] Infineon Inc., "Presece detection," Accessed : Mar. 06, 2022. [Online]. Available: https://www.infineon.com/cms/en/applications/ solutions/sensor-solutions/presence-detection/
- [25] Novelic Inc., "NOVELIC radar automotive in-cabin monitoring," Accessed : Mar. 06, 2022. [Online]. Available: https://www.novelic.com/ radar-solutions/
- [26] IEE Smart Sensing Solutions Inc., "Sensing solutions for child presence detection," Accessed : Mar. 06, 2022. [Online]. Available: https: //www.iee-sensing.com/en/automotive/safety-and-comfort/vitasense
- [27] Texas Instruments Inc., "Vehicle occupant detection reference design," Accessed : Mar. 06, 2022. [Online]. Available: https://www.ti.com/lit/ ug/tidue95a/tidue95a.pdf?ts=1646780866606
- [28] Y. Ma, Y. Zeng, and V. Jain, "Carosense: Car occupancy sensing with the ultra-wideband keyless infrastructure," in *Proc. ACM Interactive, Mobile, Wearable, Ubiquitous Technol.*, 2020, vol. 4, pp. 1–28.
- [29] G. Sikander and S. Anwar, "Driver fatigue detection systems: A review," *IEEE Trans. Intell. Transp. Syst.*, vol. 20, no. 6, pp. 2339–2352, Jun. 2019.
- [30] L. Borrelli, "Road traffic injuries," Accessed : Mar. 06, 2022. [Online]. Available: https://www.bankrate.com/insurance/car/drowsydriving-statistics/
- [31] National Highway Traffic Safety Admin., "Road traffic injuries," Accessed : Mar. 06, 2022. [Online]. Available: https://www.nhtsa.gov/risky-driving/drowsy-driving

- [32] K. T. Chui, K. F. Tsang, H. R. Chi, B. W. K. Ling, and C. K. Wu, "An accurate ecg-based transportation safety drowsiness detection scheme," *IEEE Trans. Ind. Informat.*, vol. 12, no. 4, pp. 1438–1452, Aug. 2016.
- [33] F. Wang, H. Wang, and R. Fu, "Real-time ecg-based detection of fatigue driving using sample entropy," *Entropy*, vol. 20, no. 3, 2018, Art. no. 196.
- [34] S. Murugan, J. Selvaraj, and A. Sahayadhas, "Detection and analysis: Driver state with electrocardiogram (ECG)," *Phys., Eng. Sci. Med.*, vol. 43, no. 2, pp. 525–537, 2020.
- [35] H. Lee, J. Lee, and M. Shin, "Using wearable ECG/PPG sensors for driver drowsiness detection based on distinguishable pattern of recurrence plots," *Electronics*, vol. 8, no. 2, 2019, Art. no. 192.
- [36] A. Sahayadhas, K. Sundaraj, M. Murugappan, and R. Palaniappan, "Physiological signal based detection of driver hypovigilance using higher order spectra," *Expert Syst. Appl.*, vol. 42, no. 22, pp. 8669–8677, 2015.
- [37] F. Trenta, S. Conoci, F. Rundo, and S. Battiato, "Advanced motiontracking system with multi-layers deep learning framework for innovative car-driver drowsiness monitoring," in *Proc. 14th IEEE Int. Conf. Autom. Face Gesture Recognit.*, 2019, pp. 1–5.
- [38] A. Chowdhury, R. Shankaran, M. Kavakli, and M. M. Haque, "Sensor applications and physiological features in drivers' drowsiness detection: A review," *IEEE Sensors J.*, vol. 18, no. 8, pp. 3055–3067, Apr. 2018.
- [39] S. Huang, J. Li, P. Zhang, and W. Zhang, "Detection of mental fatigue state with wearable ECG devices," *Int. J. Med. Informat.*, vol. 119, pp. 39–46, 2018.
- [40] S. Arefnezhad, A. Eichberger, M. Frühwirth, C. Kaufmann, and M. Moser, "Driver drowsiness classification using data fusion of vehiclebased measures and ECG signals," in *Proc. IEEE Int. Conf. Syst., Man, Cybern.*, 2020, pp. 451–456.
- [41] S. Hong, H. Kwon, S. H. Choi, and K. S. Park, "Intelligent system for drowsiness recognition based on ear canal electroencephalography with photoplethysmography and electrocardiography," *Inf. Sci.*, vol. 453, pp. 302–322, 2018.
- [42] N. N. Shahrndin and K. A. Sidek, "A review of ECG data acquisition for driver drowsiness detection," *Compusoft*, vol. 9, no. 7, pp. 3749–3754, 2020.
- [43] R. Roy and K. Venkatasubramanian, "EKG/ECG based driver alert system for long haul drive," *Indian J. Sci. Technol*, vol. 8, no. 19, pp. 8–13, 2015.
- [44] K. Moon, K. Hwang, E. Choi, and S. Oah, "Study on prevention of drowsiness driving using electrocardiography (LF/HF) index," J. *Korean Soc. Saf.*, vol. 30, no. 2, pp. 56–62, 2015.
- [45] A. Tjolleng *et al.*, "Classification of a driver's cognitive workload levels using artificial neural network on ECG signals," *Appl. Ergonom.*, vol. 59, pp. 326–332, 2017.
- [46] M. Babaeian and M. Mozumdar, "Driver drowsiness detection algorithms using electrocardiogram data analysis," in *Proc. IEEE 9th Annu. Comput. Commun. Workshop Conf.*, 2019, pp. 0001–0006.
- [47] M. Awais, N. Badruddin, and M. Drieberg, "A hybrid approach to detect driver drowsiness utilizing physiological signals to improve system performance and wearability," *Sensors*, vol. 17, no. 9, 2017, Art. no. 1991.
- [48] L. Wang, J. Li, and Y. Wang, "Modeling and recognition of driving fatigue state based on rr intervals of ECG data," *IEEE Access*, vol. 7, pp. 175584–175593, 2019.
- [49] J. Santamaria and K. H. Chiappa, "The EEG of drowsiness in normal adults," J. Clin. Neuriophysiol., vol. 4, no. 4, pp. 327–382, 1987.
- [50] S. Makeig and T.-P. Jung, "Tonic, phasic, and transient EEG correlates of auditory awareness in drowsiness," *Cogn. Brain Res.*, vol. 4, no. 1, pp. 15–25, 1996.
- [51] K. Fujiwara *et al.*, "Heart rate variability-based driver drowsiness detection and its validation with EEG," *IEEE Trans. Biomed. Eng.*, vol. 66, no. 6, pp. 1769–1778, Jun. 2019.
- [52] G. Li, B.-L. Lee, and W.-Y. Chung, "Smartwatch-based wearable EEG system for driver drowsiness detection," *IEEE Sensors J.*, vol. 15, no. 12, pp. 7169–7180, Dec. 2015.
- [53] T. L. da Silveira, A. J. Kozakevicius, and C. R. Rodrigues, "Automated drowsiness detection through wavelet packet analysis of a single EEG channel," *Expert Syst. Appl.*, vol. 55, pp. 559–565, 2016.



- [54] C.-S. Wei, Y.-P. Lin, Y.-T. Wang, C.-T. Lin, and T.-P. Jung, "A subject-transfer framework for obviating inter-and intra-subject variability in EEG-based drowsiness detection," *NeuroImage*, vol. 174, pp. 407–419, 2018.
- [55] Z. Gao *et al.*, "EEG-based spatio-temporal convolutional neural network for driver fatigue evaluation," *IEEE Trans. Neural Netw. Learn. Syst.*, vol. 30, no. 9, pp. 2755–2763, Sep. 2019.
- [56] G. Li and W.-Y. Chung, "A context-aware eeg headset system for early detection of driver drowsiness," *Sensors*, vol. 15, no. 8, pp. 20873–20893, 2015.
- [57] Y. Cui, Y. Xu, and D. Wu, "EEG-based driver drowsiness estimation using feature weighted episodic training," *IEEE Trans. Neural Syst. Rehabil. Eng.*, vol. 27, no. 11, pp. 2263–2273, Nov. 2019.
- [58] J. Min, P. Wang, and J. Hu, "Driver fatigue detection through multiple entropy fusion analysis in an eeg-based system," *PLoS One*, vol. 12, no. 12, 2017, Art. no. e0188756.
- [59] Y. Ma *et al.*, "Driving fatigue detection from EEG using a modified pcanet method," *Comput. Intell. Neurosci.*, vol. 2019, 2019, Art. no. 4721863.
- [60] F. Gharagozlou *et al.*, "Detecting driver mental fatigue based on EEG alpha power changes during simulated driving," *Iranian J. Public Health*, vol. 44, no. 12, 2015, Art. no. 1693.
- [61] H. Wang, C. Zhang, T. Shi, F. Wang, and S. Ma, "Real-time EEGbased detection of fatigue driving danger for accident prediction," *Int. J. Neural Syst.*, vol. 25, no. 02, 2015, Art. no. 1550002.
- [62] M. Zhu, J. Chen, H. Li, F. Liang, L. Han, and Z. Zhang, "Vehicle driver drowsiness detection method using wearable EEG based on convolution neural network," *Neural Comput. Appl.*, vol. 33, no. 20, pp. 13965–13980, 2021.
- [63] C. Wang *et al.*, "Spectral analysis of EEG during microsleep events annotated via driver monitoring system to characterize drowsiness," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 56, no. 2, pp. 1346–1356, Apr. 2020.
- [64] C.-H. Chuang, C.-S. Huang, L.-W. Ko, and C.-T. Lin, "An EEG-based perceptual function integration network for application to drowsy driving," *Knowl.-Based Syst.*, vol. 80, pp. 143–152, 2015.
- [65] I. Stancin, M. Cifrek, and A. Jovic, "A review of EEG signal features and their application in driver drowsiness detection systems," *Sensors*, vol. 21, no. 11, 2021, Art. no. 3786.
- [66] H. Wang, A. Dragomir, N. I. Abbasi, J. Li, N. V. Thakor, and A. Bezerianos, "A novel real-time driving fatigue detection system based on wireless dry EEG," *Cogn. Neurodynamics*, vol. 12, no. 4, pp. 365–376, 2018.
- [67] L. Boon-Leng, L. Dae-Seok, and L. Boon-Giin, "Mobile-based wearable-type of driver fatigue detection by GSR and EMG," in *Proc. IEEE Region 10 Conf.*, 2015, pp. 1–4.
- [68] M. Mahmoodi and A. Nahvi, "Driver drowsiness detection based on classification of surface electromyography features in a driving simulator," *Proc. Inst. Mech. Eng., Part H: J. Eng. Med.*, vol. 233, no. 4, pp. 395–406, 2019.
- [69] R. Fu and H. Wang, "Detection of driving fatigue by using noncontact EMG and ECG signals measurement system," *Int. J. Neural Syst.*, vol. 24, no. 03, 2014, Art. no. 1450006.
- [70] C. Katsis, N. Ntouvas, C. Bafas, and D. Fotiadis, "Assessment of muscle fatigue during driving using surface EMG," in *Proc. IASTED Int. Conf. Biomed. Eng.*, 2004, pp. 259–262.
- [71] Y. Fan, F. Gu, J. Wang, J. Wang, K. Lu, and J. Niu, "Safedriving: An effective abnormal driving behavior detection system based on emg signals," *IEEE Internet Things J.*, early access, 2021, doi: 10.1109/JIOT.2021.3135512.
- [72] N. Ab Rahman, M. Mustafa, N. Sulaiman, R. Samad, and N. Abdullah, "EMG signal segmentation to predict driver's vigilance state," in *Human-Centered Technol. Better Tomorrow*, Berlin, Germany: Springer, 2022, pp. 29–42.
- [73] R. S. Concepcion *et al.*, "Alertness and mental fatigue classification using computational intelligence in an electrocardiography and electromyography system with off-body area network," in *World Congr. Eng. Technol.; Innov. Sustainability*, Berlin, Germany: Springer, 2018, pp. 153–169.
- [74] E. J. C. Nacpil, Z. Wang, and K. Nakano, "Application of physiological sensors for personalization in semi-autonomous driving: A review," *IEEE Sensors J.*, vol. 21, no. 18, pp. 19662–19674, Sep. 2021.

- [75] M.-S. Qi, W.-J. Yang, P. Xie, Z.-J. Liu, Y.-Y. Zhang, and S.-C. Cheng, "Driver fatigue assessment based on the feature fusion and transfer learning of EEG and EMG," in *Proc. Chin. Automat. Congr.*, 2018, pp. 1314–1317.
- [76] F. Naim, M. Mustafa, N. Sulaiman, and N. A. A. Rahman, "The study of time domain features of EMG signals for detecting driver's drowsiness," in *Recent Trends Mechatronics Towards Industry 4.0.* Berlin, Germany: Springer, 2022, pp. 427–438.
- [77] K. Li, Y. Gong, and Z. Ren, "A fatigue driving detection algorithm based on facial multi-feature fusion," *IEEE Access*, vol. 8, pp. 101244–101259, 2020.
- [78] W. Liu, J. Qian, Z. Yao, X. Jiao, and J. Pan, "Convolutional two-stream network using multi-facial feature fusion for driver fatigue detection," *Future Internet*, vol. 11, no. 5, 2019, Art. no. 115.
- [79] Z. Liu, Y. Peng, and W. Hu, "Driver fatigue detection based on deeply-learned facial expression representation," J. Vis. Commun. Image Representation, vol. 71, 2020, Art. no. 102723.
- [80] L. Chen, G. Xin, Y. Liu, and J. Huang, "Driver fatigue detection based on facial key points and LSTM," *Secur. Commun. Netw.*, vol. 2021, 2021, Art. no. 5383573.
- [81] N. Irtija, M. Sami, and M. A. R. Ahad, "Fatigue detection using facial landmarks," in *Proc. Int. Symp. Affect. Sci. Eng.*, 2018, pp. 1–6.
- [82] G. Sikander and S. Anwar, "A novel machine vision-based 3D facial action unit identification for fatigue detection," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 5, pp. 2730–2740, May 2020.
- [83] S. A. Khan, S. Hussain, S. Xiaoming, and S. Yang, "An effective framework for driver fatigue recognition based on intelligent facial expressions analysis," *IEEE Access*, vol. 6, pp. 67459–67468, 2018.
- [84] W. H. Gu, Y. Zhu, X. D. Chen, L. F. He, and B. B. Zheng, "Hierarchical CNN-based real-time fatigue detection system by visual-based technologies using MSP model," *IET Image Process.*, vol. 12, no. 12, pp. 2319–2329, 2018.
- [85] S. Dey, S. A. Chowdhury, S. Sultana, M. A. Hossain, M. Dey, and S. K. Das, "Real time driver fatigue detection based on facial behaviour along with machine learning approaches," in *Proc. IEEE Int. Conf. Signal Process., Inf., Commun. Syst.*, 2019, pp. 135–140.
- [86] X. Li, J. Xia, L. Cao, G. Zhang, and X. Feng, "Driver fatigue detection based on convolutional neural network and face alignment for edge computing device," *Proc. Inst. Mech. Eng., Part D: J. Automobile Eng.*, vol. 235, no. 10–11, pp. 2699–2711, 2021.
- [87] K. Li, S. Wang, C. Du, Y. Huang, X. Feng, and F. Zhou, "Accurate fatigue detection based on multiple facial morphological features," *J. Sensors*, vol. 2019, 2019, Art. no. 7934516.
- [88] S. Abtahi, M. Omidyeganeh, S. Shirmohammadi, and B. Hariri, "Yawdd: A yawning detection dataset," in *Proc. 5th ACM Multimedia Syst. Conf.*, 2014, pp. 24–28.
- [89] F. Dornaika, J. Reta, I. Arganda-Carreras, and A. Moujahid, "Driver drowsiness detection in facial images," in *Proc. 8th Int. Conf. Image Process. Theory, Tools, Appl.*, 2018, pp. 1–6.
- [90] A. Moujahid, F. Dornaika, I. Arganda-Carreras, and J. Reta, "Efficient and compact face descriptor for driver drowsiness detection," *Expert Syst. Appl.*, vol. 168, 2021, Art. no. 114334.
- [91] S. Junaedi and H. Akbar, "Driver drowsiness detection based on face feature and perclos," in *Proc. J. Phys.: Conf. Ser.*, 2018, vol. 1090, Art. no. 012037.
- [92] J. Xing, G. Fang, J. Zhong, and J. Li, "Application of face recognition based on CNN in fatigue driving detection," in *Proc. Int. Conf. Artif. Intell. Adv. Manuf.*, 2019, pp. 1–5.
- [93] G. Du, T. Li, C. Li, P. X. Liu, and D. Li, "Vision-based fatigue driving recognition method integrating heart rate and facial features," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 5, pp. 3089–3100, May 2020.
- [94] M. Knapik and B. Cyganek, "Driver's fatigue recognition based on yawn detection in thermal images," *Neurocomputing*, vol. 338, pp. 274–292, 2019.
- [95] L. Zhao, Z. Wang, X. Wang, and Q. Liu, "Driver drowsiness detection using facial dynamic fusion information and a DBN," *IET Intell. Transport Syst.*, vol. 12, no. 2, pp. 127–133, 2018.
- [96] H. Tao, G. Zhang, Y. Zhao, and Y. Zhou, "Real-time driver fatigue detection based on face alignment," in *Proc. 9th Int. Conf. Digit. Image Process.*, 2017, vol. 10420, pp. 6–11.
- [97] Y. Wang, R. Huang, and L. Guo, "Eye gaze pattern analysis for fatigue detection based on GP-BCNN with ESM," *Pattern Recognit. Lett.*, vol. 123, pp. 61–74, 2019.

- [98] F. Zhang, J. Su, L. Geng, and Z. Xiao, "Driver fatigue detection based on eye state recognition," in *Proc. Int. Conf. Mach. Vis. Inf. Technol.*, 2017, pp. 105–110.
- [99] S. Sathasivam, A. K. Mahamad, S. Saon, A. Sidek, M. M. Som, and H. A. Ameen, "Drowsiness detection system using eye aspect ratio technique," in *Proc. IEEE Student Conf. Res. Develop.*, 2020, pp. 448–452.
- [100] J. Xu, J. Min, and J. Hu, "Real-time eye tracking for the assessment of driver fatigue," *Healthcare Technol. Lett.*, vol. 5, no. 2, pp. 54–58, 2018.
- [101] B. Mandal, L. Li, G. S. Wang, and J. Lin, "Towards detection of bus driver fatigue based on robust visual analysis of eye state," *IEEE Trans. Intell. Transp. Syst.*, vol. 18, no. 3, pp. 545–557, Mar. 2017.
- [102] L. Lin et al., "Driver fatigue detection based on eye state," Technol. Health Care, vol. 23, no. s2, pp. S453–S463, 2015.
- [103] M. A. M. Bhoyar and S. Sawalkar, "Implementation on visual analysis of eye state using image processing for driver fatigue detection," *Int. Res. J. Eng. Technol.*, vol. 6, no. 4, pp. 4340–4346, 2019.
- [104] W. G. Aguilar, J. I. Estrella, W. López, and V. Abad, "Driver fatigue detection based on real-time eye gaze pattern analysis," in *Proc. Int. Conf. Intell. Robot. Appl.*, 2017, pp. 683–694.
- [105] P. Yan, Y. Sun, Z. Li, J. Zou, and D. Hong, "Driver fatigue detection system based on colored and infrared eye features fusion," *Comput., Mater. Continua*, vol. 63, no. 3, pp. 1563–1574, 2020.
- [106] P. Chen, "Research on driver fatigue detection strategy based on human eye state," in *Proc. Chin. Automat. Congr.*, 2017, pp. 619–623.
- [107] R. Galindo et al., "Landmark based eye ratio estimation for driver fatigue detection," in Proc. Int. Conf. Intell. Robot. Appl., 2019, pp. 565–576.
- [108] Q. Zhuang, Z. Kehua, J. Wang, and Q. Chen, "Driver fatigue detection method based on eye states with pupil and iris segmentation," *IEEE Access*, vol. 8, pp. 173440–173449, 2020.
- [109] M. C. Catalbas, T. Cegovnik, J. Sodnik, and A. Gulten, "Driver fatigue detection based on saccadic eye movements," in *Proc. 10th Int. Conf. Elect. Electron. Eng.*, 2017, pp. 913–917.
- [110] X. Hu and G. Lodewijks, "Exploration of the effects of task-related fatigue on eye-motion features and its value in improving driver fatigue-related technology," *Transp. Res. Part F: Traffic Psychol. Behav.*, vol. 80, pp. 150–171, 2021.
- [111] S. Park, F. Pan, S. Kang, and C. D. Yoo, "Driver drowsiness detection system based on feature representation learning using various deep networks," in *Proc. Asian Conf. Comput. Vis.*, 2016, pp. 154–164.
- [112] National Tsing Hua University, "Nthuddd video dataset," Accessed : Mar. 06, 2022. [Online]. Available: http://cv.cs.nthu.edu.tw/php/ callforpaper/2016\_ACCVworkshop/
- [113] E. Murphy-Chutorian and M. M. Trivedi, "Head pose estimation and augmented reality tracking: An integrated system and evaluation for monitoring driver awareness," *IEEE Trans. Intell. Transp. Syst.*, vol. 11, no. 2, pp. 300–311, Jun. 2010.
- [114] A. Tawari, S. Martin, and M. M. Trivedi, "Continuous head movement estimator for driver assistance: Issues, algorithms, and on-road evaluations," *IEEE Trans. Intell. Transp. Syst.*, vol. 15, no. 2, pp. 818–830, Apr. 2014.
- [115] S. Martin, A. Tawari, and M. M. Trivedi, "Monitoring head dynamics for driver assistance systems: A multi-perspective approach," in *Proc. 16th Int. IEEE Conf. Intell. Transp. Syst.*, 2013, pp. 2286–2291.
- [116] E. Vural, M. Cetin, A. Ercil, G. Littlewort, M. Bartlett, and J. Movellan, "Drowsy driver detection through facial movement analysis," in *Proc. Int. Workshop Hum.-Comput. Interact.*, 2007, pp. 6–18.
- [117] C. Craye and F. Karray, "Driver distraction detection and recognition using RGB-D sensor," 2015, arXiv:1502.00250. [Online]. Available: https://www.researchgate.net/profile/Fakhri-Karray/publication/2717-71395\_Driver\_distraction\_detection\_and\_recognition\_using\_RGB-D\_sensor/links/54e3765b0cf282dbed6bed4f/Driver-distraction-detection-and-recognition-using-RGB-D-sensor.pdf
- [118] Q. Wang, J. Yang, M. Ren, and Y. Zheng, "Driver fatigue detection: A survey," in *Proc. 6th World Congr. Intell. Control, Automat.*, 2006, vol. 2, pp. 8587–8591.
- [119] A. Mittal, K. Kumar, S. Dhamija, and M. Kaur, "Head movementbased driver drowsiness detection: A review of state-of-art techniques," in *Proc. IEEE Int. Conf. Eng. Technol.*, 2016, pp. 903–908.
- [120] J. Hu, L. Xu, X. He, and W. Meng, "Abnormal driving detection based on normalized driving behavior," *IEEE Trans. Veh. Technol.*, vol. 66, no. 8, pp. 6645–6652, Aug. 2017.

- [121] W. Tansakul and P. Tangamchit, "Fatigue driver detection system using a combination of blinking rate and driving inactivity," *J. Automat. Control Eng.*, vol. 4, no. 1, pp. 33–39, 2016.
- [122] R. Gupta, K. Aman, N. Shiva, and Y. Singh, "An improved fatigue detection system based on behavioral characteristics of driver," in *Proc. 2nd IEEE Int. Conf. Intell. Transp. Eng.*, 2017, pp. 227–230.
- [123] M. Hou, M. Wang, W. Zhao, Q. Ni, Z. Cai, and X. Kong, "A lightweight framework for abnormal driving behavior detection," *Comput. Commun.*, vol. 184, pp. 128–136, 2022.
- [124] M. Shahverdy, M. Fathy, R. Berangi, and M. Sabokrou, "Driver behavior detection and classification using deep convolutional neural networks," *Expert Syst. Appl.*, vol. 149, 2020, Art. no. 113240.
- [125] R. Chhabra, S. Verma, and C. R. Krishna, "A survey on driver behavior detection techniques for intelligent transportation systems," in *Proc. 7th Int. Conf. Cloud Comput., Data Sci. Eng.-Confluence*, 2017, pp. 36–41.
- [126] T. K. Chan, C. S. Chin, H. Chen, and X. Zhong, "A comprehensive review of driver behavior analysis utilizing smartphones," *IEEE Trans. Intell. Transp. Syst.*, vol. 21, no. 10, pp. 4444–4475, Oct. 2020.
- [127] B. Shi *et al.*, "Evaluating driving styles by normalizing driving behavior based on personalized driver modeling," *IEEE Trans. Syst., Man, Cybern. Syst.*, vol. 45, no. 12, pp. 1502–1508, Dec. 2015.
- [128] S. Kaplan, M. A. Guvensan, A. G. Yavuz, and Y. Karalurt, "Driver behavior analysis for safe driving: A survey," *IEEE Trans. Intell. Transp. Syst.*, vol. 16, no. 6, pp. 3017–3032, Dec. 2015.
- [129] M. Wollmer *et al.*, "Online driver distraction detection using long short-term memory," *IEEE Trans. Intell. Transp. Syst.*, vol. 12, no. 2, pp. 574–582, Jun. 2011.
- [130] F. Tango and M. Botta, "Real-time detection system of driver distraction using machine learning," *IEEE Trans. Intell. Transp. Syst.*, vol. 14, no. 2, pp. 894–905, Jun. 2013.
- [131] A. Sathyanarayana, S. Nageswaren, H. Ghasemzadeh, R. Jafari, and J. H. Hansen, "Body sensor networks for driver distraction identification," in *Proc. IEEE Int. Conf. Veh. Electron. Saf.*, 2008, pp. 120–125.
- [132] T. Liu, Y. Yang, G.-B. Huang, Y. K. Yeo, and Z. Lin, "Driver distraction detection using semi-supervised machine learning," *IEEE Trans. Intell. Transp. Syst.*, vol. 17, no. 4, pp. 1108–1120, Apr. 2016.
- [133] FaceLAB, "Facelab eye tracker," Accessed : Mar. 06, 2022. [Online]. Available: https://www.eyecomtec.com/3132-faceLAB
- [134] M. Kutila, M. Jokela, G. Markkula, and M. R. Rué, "Driver distraction detection with a camera vision system," in *Proc. IEEE Int. Conf. Image Process.*, 2007, vol. 6, pp. VI–201.
- [135] Y. Liang, M. L. Reyes, and J. D. Lee, "Real-time detection of driver cognitive distraction using support vector machines," *IEEE Trans. Intell. Transp. Syst.*, vol. 8, no. 2, pp. 340–350, Jun. 2007.
- [136] Z. Zhao *et al.*, "Driver distraction detection method based on continuous head pose estimation," *Comput. Intell. Neurosci.*, vol. 2020, 2020, Art. no. 9606908.
- [137] M. Ye, W. Zhang, P. Cao, and K. Liu, "Driver fatigue detection based on residual channel attention network and head pose estimation," *Appl. Sci.*, vol. 11, no. 19, 2021, Art. no. 9195.
- [138] T. Singh, M. Mohadikar, S. Gite, S. Patil, B. Pradhan, and A. Alamri, "Attention span prediction using head-pose estimation with deep neural networks," *IEEE Access*, vol. 9, pp. 142632–142643, 2021.
- [139] A. Ragab, C. Craye, M. S. Kamel, and F. Karray, "A visual-based driver distraction recognition and detection using random forest," in *Proc. Int. Conf. Image Anal. Recognit.*, 2014, pp. 256–265.
- [140] A. Fernández, R. Usamentiaga, J. L. Carús, and R. Casado, "Driver distraction using visual-based sensors and algorithms," *Sensors*, vol. 16, no. 11, 2016, Art. no. 1805.
- [141] Y. Liang and J. D. Lee, "Driver cognitive distraction detection using eye movements," in *Passive Eye Monit.*, Berlin, Germany: Springer, 2008, pp. 285–300.
- [142] M. Rezaei and R. Klette, "3D cascade of classifiers for open and closed eye detection in driver distraction monitoring," in *Proc. Int. Conf. Comput. Anal. Images Patterns*, 2011, pp. 171–179.
- [143] M.-H. Sigari, M. Fathy, and M. Soryani, "A driver face monitoring system for fatigue and distraction detection," *Int. J. Veh. Technol.*, vol. 2013, 2013, Art. no. 263983.
- [144] M.-H. Sigari, M.-R. Pourshahabi, M. Soryani, and M. Fathy, "A review on driver face monitoring systems for fatigue and distraction detection," *Int. J. Adv. Sci. Technol.*, vol. 64, pp. 73–100, 2014.

Vehicular Technology

- [145] W. Li, J. Huang, G. Xie, F. Karray, and R. Li, "A survey on visionbased driver distraction analysis," J. Syst. Architecture, vol. 121, 2021, Art. no. 102319.
- [146] A. Kashevnik, R. Shchedrin, C. Kaiser, and A. Stocker, "Driver distraction detection methods: A literature review and framework," *IEEE Access*, vol. 9, pp. 60063–60076, 2021.
- [147] S. Zepf, J. Hernandez, A. Schmitt, W. Minker, and R. W. Picard, "Driver emotion recognition for intelligent vehicles: A survey," ACM Comput. Surv., vol. 53, no. 3, pp. 1–30, 2020.
- [148] G. Du, Z. Wang, B. Gao, S. Mumtaz, K. M. Abualnaja, and C. Du, "A convolution bidirectional long short-term memory neural network for driver emotion recognition," *IEEE Trans. Intell. Transp. Syst.*, vol. 22, no. 7, pp. 4570–4578, Jul. 2021.
- [149] S. D. Regani, Q. Xu, B. Wang, M. Wu, and K. J. R. Liu, "Driver authentication for smart car using wireless sensing," *IEEE Internet Things J.*, vol. 7, no. 3, pp. 2235–2246, Mar. 2020.
- [150] A. Burton *et al.*, "Driver identification and authentication with active behavior modeling," in *Proc. 12th Int. Conf. Netw. Serv. Manage.*, 2016, pp. 388–393.
- [151] Y. Xun, J. Liu, N. Kato, Y. Fang, and Y. Zhang, "Automobile driver fingerprinting: A new machine learning based authentication scheme," *IEEE Trans. Ind. Informat.*, vol. 16, no. 2, pp. 1417–1426, Feb. 2020.
- [152] E. Derman and A. A. Salah, "Continuous real-time vehicle driver authentication using convolutional neural network based face recognition," in *Proc. 13th IEEE Int. Conf. Autom. Face Gesture Recognit.*, 2018, pp. 577–584.
- [153] P. Nissimagoudar and A. Nandi, "Precision enhancement of driver assistant system using EEG based driver consciousness analysis & classification," *Comput. Netw. Appl. Tools Perform. Manage.*, Berlin, Germany: Springer-Verlag, Oct. 2020, pp. 247–257.
- [154] M. Gromer, D. Salb, T. Walzer, N. Madrid, and R. Seepold, "ECG sensor for detection of driver's drowsiness," *Procedia Comput. Sci.*, vol. 159, pp. 1938–1946, Jan. 2019.
- [155] K. Chui, K. Tsang, H. Chi, C. Wu, and B. Ling, "Electrocardiogram based classifier for driver drowsiness detection," in *Proc. IEEE 13th Int. Conf. Ind. Informat.*, 2015, pp. 600–603.
- [156] S. Hu, R. Bowlds, Y. Gu, and X. Yu, "Pulse wave sensor for nonintrusive driver's drowsiness detection," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2009, pp. 2312–2315.
- [157] J.-K. Park *et al.*, "Noncontact RF vital sign sensor for continuous monitoring of driver status," *IEEE Trans. Biomed. Circuits Syst.*, vol. 13, no. 3, pp. 493–502, Jun. 2019.
- [158] P. Vetter, L. Leicht, S. Leonhardt, and D. Teichmann, "Integration of an electromagnetic coupled sensor into a driver seat for vital sign monitoring: Initial insight," in *Proc. IEEE Int. Conf. Veh. Electron. Saf.*, 2017, pp. 185–190.
- [159] M. Hussain, A. Akbilek, F. Pfeiffer, and B. Napholz, "In-vehicle breathing rate monitoring based on WiFi signals," in *Proc. 50th Eur. Microw. Conf.*, 2021, pp. 292–295.
- [160] F. Wang, F. Zhang, C. Wu, B. Wang, and K. J. R. Liu, "Respiration tracking for people counting and recognition," *IEEE Internet Things J.*, vol. 7, no. 6, pp. 5233–5245, Jun. 2020.
- [161] F. Zhang et al., "SMARS: Sleep monitoring via ambient radio signals," *IEEE Trans. Mobile Comput.*, vol. 20, no. 1, pp. 217–231, Jan. 2021.
- [162] C. Chen et al., "TR-BREATH: Time-reversal breathing rate estimation and detection," *IEEE Trans. Biomed. Eng.*, vol. 65, no. 3, pp. 489–501, Mar. 2018.
- [163] F. Wang, F. Zhang, C. Wu, B. Wang, and K. J. R. Liu, "ViMo: Multiperson vital sign monitoring using commodity millimeter-wave radio," *IEEE Internet Things J.*, vol. 8, no. 3, pp. 1294–1307, Feb. 2021.
- [164] J. Jaworek-Korjakowska, A. Kostuch, and P. Skruch, "SAFESO: Interpretable and explainable deep learning approach for seat occupancy classification in vehicle interior," in *Proc. IEEE/CVF Conf. Comput. Vis. Pattern Recognit.*, 2021, pp. 103–112.
- [165] United States Department of Transportation, "Seat belts," Accessed : Mar. 06, 2022. [Online]. Available: https://www.nhtsa.gov/riskydriving/seat-belts
- [166] Euro NCAP, "NCAP Euro 2025 Roadmap," Accessed : Mar. 06, 2022. [Online]. Available: https://cdn.euroncap.com/media/30700/ euroncap-roadmap-2025-v4.pdf

- [167] A. Rakshitha *et al.*, "A child-left behind warning system using capacitive sensing principle," *Int. J. Innov. Res. Comput. Commun. Eng*, vol. 5, no. 5, pp. 9526–9531, 2017.
- [168] F. R. Rashidi and I. H. Muhamad, "Vehicle's interior movement detection and notification system," in *Proc. Recent Adv. Autom. Control, Model. Simul.*, 2013, pp. 139–144.
- [169] M. Hansjürg, R. Martin, and A. Rolf, "Presence detector and its application," U.S. Patent 6, 486, 778, Nov. 26, 2002.
- [170] P. Zappi, E. Farella, and L. Benini, "Tracking motion direction and distance with pyroelectric ir sensors," *IEEE Sensors J.*, vol. 10, no. 9, pp. 1486–1494, Sep. 2010.
- [171] J. L. Wilson, "Automotive WiFi availability in dynamic urban canyon environments," *Navigation: J. Inst. Navigation*, vol. 63, no. 2, pp. 161–172, 2016.
- [172] L. Bell, "Cheap cars with WiFi capability," Accessed : Mar. 06, 2022. [Online]. Available: https://www.autobytel.com/car-buyingguides/features/cheap-cars-with-wifi-capability-130477/
- [173] H. A. Research, "Cars with Wi-Fi: Everything you need to know," Accessed : Mar. 06, 2022. [Online]. Available: https://www.caranddriver. com/research/a32814112/cars-with-wifi/
- [174] C. Chen, Y. Chen, Y. Han, H.-Q. Lai, and K. J. R. Liu, "Achieving centimeter-accuracy indoor localization on wifi platforms: A frequency hopping approach," *IEEE Internet Things J.*, vol. 4, no. 1, pp. 111–121, Feb. 2017.
- [175] Y. Xie, Z. Li, and M. Li, "Precise power delay profiling with commodity Wi-Fi," *IEEE Trans. Mobile Comput.*, vol. 18, no. 6, pp. 1342–1355, Jun. 2019.
- [176] C. Fan, Y. Wang, and C. Huang, "Heterogeneous information fusion and visualization for a large-scale intelligent video surveillance system," *IEEE Trans. Syst., Man, Cybern.: Syst.*, vol. 47, no. 4, pp. 593–604, Apr. 2017.
- [177] F. Erlik Nowruzi, W. A. El Ahmar, R. Laganiere, and A. H. Ghods, "In-vehicle occupancy detection with convolutional networks on thermal images," in *Proc. IEEE/CVF Conf. Comput. Vis. Pattern Recognit. Workshops*, 2019, pp. 941–948.
- [178] S. D. D. Cruz, O. Wasenmuller, H.-P. Beise, T. Stifter, and D. Stricker, "Sviro: Synthetic vehicle interior rear seat occupancy dataset and benchmark," in *Proc. IEEE/CVF Winter Conf. Appl. Comput. Vis.*, 2020, pp. 973–982.
- [179] K. He, X. Zhang, S. Ren, and J. Sun, "Deep residual learning for image recognition," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit.*, 2016, pp. 770–778.
- [180] G. Huang, Z. Liu, L. Van Der Maaten, and K. Q. Weinberger, "Densely connected convolutional networks," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit.*, 2017, pp. 4700–4708.
- [181] M. Tan and Q. E. Le, "Rethinking model scaling for convolutional neural networks," in *Proc. Int. Conf. Mach. Learn.*, 2019, pp. 6105–6014.
- [182] X. Zeng, B. Chen, and M. Yang, "Doa estimation for low angle targets using time reversal in frequency domain model," in *Proc. IEEE Radar Conf.*, 2018, pp. 1323–1327.
  [183] X. Zeng, M. Yang, B. Chen, and Y. Jin, "Estimation of di-
- [183] X. Zeng, M. Yang, B. Chen, and Y. Jin, "Estimation of direction of arrival by time reversal for low-angle targets," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 54, no. 6, pp. 2675–2694, Dec. 2018.
- [184] X. Zeng, M. Yang, B. Chen, and Y. Jin, "Low angle direction of arrival estimation by time reversal," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process.*, 2017, pp. 3161–3165.
- [185] X. Zeng, F. Zhang, B. Wang, and K. J. R. Liu, "High accuracy tracking of targets using massive MIMO," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process.*, 2021, pp. 4435–4439.
- [186] X. Zeng, F. Zhang, B. Wang, and K. J. R. Liu, "Massive MIMO for high-accuracy target localization and tracking," *IEEE Internet Things J.*, vol. 8, no. 12, pp. 10131–10145, Jun. 2021.
- [187] X. Zeng, F. Zhang, B. Wang, and K. J. R. Liu, "Radio frequency based direction sensing using massive MIMO," *IEEE Access*, vol. 8, pp. 26827–26838, 2020.
- [188] Federal Communications Commission, "FCC seeks to enable state-of-the-art radar sensors in 60 GHz band," Accessed : Mar. 06, 2022. [Online]. Available: https://www.federalregister.gov/ documents/2021/08/19/2021-16637/fcc-seeks-to-enable-state-ofthe-art-radar-sensors-in-60-ghz-band

- [189] IEE Smart Sensing Solutions. Inc., "Our sensors make your life better, safer and more comfortable," Accessed : Mar. 06, 2022. [Online]. Available: https://iee-sensing.com/in-vehicle-radar-technologyvitasense-certified-in-the-us-and-canada/
- [190] IEE Smart Sensing Solutions. Inc., "Safety & comfort in-vehicle sensing solutions," Accessed : Mar. 06, 2022. [Online]. Available: https://iee-sensing.com/automotive/safety-and-comfort/
- [191] Printed Electronics, "IEE launches vitasense," Accessed : Mar. 06, 2022. [Online]. Available: https://www.printedelectronicsnow. com/contents/view\_breaking-news/2021-04-26/iee-launchesvitasense/
- [192] L. Wang, H. Wang, and X. Jiang, "A new method to detect driver fatigue based on EMG and ECG collected by portable noncontact sensors," *Promet-Traffic Transp.*, vol. 29, no. 5, pp. 479–488, 2017.
- [193] J. Kim and M. Shin, "Utilizing hrv-derived respiration measures for driver drowsiness detection," *Electronics*, vol. 8, no. 6, 2019, Art. no. 669.
- [194] H. Al-Libawy, A. Al-Ataby, W. Al-Nuaimy, and M. A. Al-Taee, "HRV-based operator fatigue analysis and classification using wearable sensors," in *Proc. 13th Int. Multi- Conf. Syst., Signals, Devices*, 2016, pp. 268–273.
- [195] S. Cateni, V. Colla, M. Vannucci, and A. Borselli, "Fuzzy inference systems applied to image classification in the industrial field," in *Fuzzy Inference System: Theory and Applications.*, London, U.K.: InTech, 2012, [Online]. Available: https://www.intechopen.com/books/1918
- [196] Z. J. Viharos and K. B. Kis, "Survey on neuro-fuzzy systems and their applications in technical diagnostics and measurement," *Measurement*, vol. 67, pp. 126–136, 2015.
- [197] S. Arefnezhad, S. Samiee, A. Eichberger, and A. Nahvi, "Driver drowsiness detection based on steering wheel data applying adaptive neuro-fuzzy feature selection," *Sensors*, vol. 19, no. 4, 2019, Art. no. 943.
- [198] J. Redmon and A. Farhadi, "Yolov3: An incremental improvement," 2018, arXiv:1804.02767.
- [199] D. E. King, "Dlib-ml: A machine learning toolkit," J. Mach. Learn. Res., vol. 10, pp. 1755–1758, 2009.
- [200] S. Lin, Y. Tan, P. Chua, L. Tey, and C. Ang, "Perclos threshold for drowsiness detection during real driving," J. Vis., vol. 12, pp. 546–546, Aug. 2012.
- [201] C.-H. Weng, Y.-H. Lai, and S.-H. Lai, "Driver drowsiness detection via a hierarchical temporal deep belief network," in *Proc. Asian Conf. Comput. Vis.*, 2016, pp. 117–133.
- [202] A. Tawari and M. M. Trivedi, "Robust and continuous estimation of driver gaze zone by dynamic analysis of multiple face videos," in *Proc. IEEE Intell. Veh. Symp.*, 2014, pp. 344–349.
- [203] C. Blaschke, F. Breyer, B. Färber, J. Freyer, and R. Limbacher, "Driver distraction based lane-keeping assistance," *Transp. Res. Part F: Traffic Psychol. Behav.*, vol. 12, no. 4, pp. 288–299, 2009.
- [204] "Controller area network," Accessed : Mar. 06, 2022. [Online]. Available: https://en.wikipedia.org/wiki/CAN\_bus
- [205] J. D. Lee, K. L. Young, and M. A. Regan, "Defining driver distraction," *Driver Distraction: Theory, Effects, Mitigation*, vol. 13, no. 4, pp. 31–40, 2008.
- [206] M. A. Regan, C. Hallett, and C. P. Gordon, "Driver distraction and driver inattention: Definition, relationship and taxonomy," *Accident Anal. Prevention*, vol. 43, no. 5, pp. 1771–1781, 2011.
- [207] M. A. Regan, J. D. Lee, and K. Young, Driver Distraction: Theory, Effects, and Mitigation. Boca Raton, FL, USA: CRC Press, 2008.
- [208] G. Underwood, P. Chapman, S. Wright, and D. Crundall, "Anger while driving," *Transp. Res. Part F: Traffic Psychol. Behav.*, vol. 2, no. 1, pp. 55–68, 1999.
- [209] S. Fairclough, A. Tattersall, and K. Houston, "Anxiety and performance in the british driving test," *Transp. Res. Part F: Traffic Psychol. Behav.*, vol. 9, pp. 43–52, Jan. 2006.
- [210] M. Jeon, "Don't cry while you're driving: Sad driving is as bad as angry driving," *Int. J. Hum.-Comput. Interact.*, vol. 32, no. 10, pp. 777–790, 2016.
- [211] J. Siegal, "Tesla video play feture," Accessed : Mar. 06, 2022. [Online]. Available: https://bgr.com/tech/tesla-wont-let-you-play-videogames-in-moving-cars-anymore/
- [212] L. Yekhshatyan and J. D. Lee, "Changes in the correlation between eye and steering movements indicate driver distraction," *IEEE Trans. Intell. Transp. Syst.*, vol. 14, no. 1, pp. 136–145, Mar. 2013.

- [213] J. M. Mase, P. Chapman, G. P. Figueredo, and M. T. Torres, "A hybrid deep learning approach for driver distraction detection," in *Proc. Int. Conf. Inf. Commun. Technol. Convergence*, 2020, pp. 1–6.
- [214] K. Kircher and C. Ahlström, "The driver distraction detection algorithm attend," in *Driver Distraction Inattention*, Boca Raton, FL, USA: CRC Press, 2017, pp. 327–348.
- [215] C. Ahlström, G. Georgoulas, and K. Kircher, "Towards a contextdependent multi-buffer driver distraction detection algorithm," *IEEE Trans. Intell. Transp. Syst.*, vol. 23, no. 5, pp. 4778–4790, May 2022.
- [216] C. Ahlstrom, K. Kircher, and A. Kircher, "A gaze-based driver distraction warning system and its effect on visual behavior," *IEEE Trans. Intell. Transp. Syst.*, vol. 14, no. 2, pp. 965–973, Jun. 2013.
- [217] M. S. Majdi, S. Ram, J. T. Gill, and J. J. Rodríguez, "Drivenet: Convolutional network for driver distraction detection," in *Proc. IEEE Southwest Symp. Image Anal. Interpretation*, 2018, pp. 1–4.
- [218] Kaggle Inc., "State farm distracted driver detection," Accessed : Mar. 06, 2022. [Online]. Available: https://www.kaggle.com/c/state-farmdistracted-driver-detection
- [219] A. Aksjonov, P. Nedoma, V. Vodovozov, E. Petlenkov, and M. Herrmann, "Detection and evaluation of driver distraction using machine learning and fuzzy logic," *IEEE Trans. Intell. Transp. Syst.*, vol. 20, no. 6, pp. 2048–2059, Jun. 2019.
- [220] D. K. Jain, R. Jain, X. Lan, Y. Upadhyay, and A. Thareja, "Driver distraction detection using capsule network," *Neural Comput. Appl.*, vol. 33, no. 11, pp. 6183–6196, 2021.
- [221] X. Zhu, Z. Lei, X. Liu, H. Shi, and S. Z. Li, "Face alignment across large poses: A. 3D solution," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit.*, 2016, pp. 146–155.
- [222] M. Koestinger, P. Wohlhart, P. M. Roth, and H. Bischof, "Annotated facial landmarks in the wild: A large-scale, real-world database for facial landmark localization," in *Proc. IEEE Int. Conf. Comput. Vis. Workshops*, 2011, pp. 2144–2151.
- [223] C. Hari and P. Sankaran, "Driver distraction analysis using face pose cues," *Expert Syst. Appl.*, vol. 179, 2021, Art. no. 115036.
- [224] H. V. Koay, J. H. Chuah, C.-O. Chow, Y.-L. Chang, and B. Rudrusamy, "Optimally-weighted image-pose approach (owipa) for distracted driver detection and classification," *Sensors*, vol. 21, no. 14, 2021, Art. no. 4837.
- [225] K. Kircher, C. Ahlstrom, and A. Kircher, "Comparison of two eyegaze based real-time driver distraction detection algorithms in a small-scale field operational test," in *Proc. 5th Int. Symp. Hum. Factors Driver Assessment, Training, Veh. Des.*, 2009, pp. 16–23.
- [226] A. Azman, Q. Meng, and E. Edirisinghe, "Non intrusive physiological measurement for driver cognitive distraction detection: Eye and mouth movements," in *Proc. 3rd Int. Conf. Adv. Comput. Theory Eng.*, 2010, vol. 3, pp. V3–595.
- [227] W. Kim, H.-K. Choi, B.-T. Jang, and J. Lim, "Driver distraction detection using single convolutional neural network," in *Proc. Int. Conf. Inf. Commun. Technol. Convergence*, 2017, pp. 1203–1205.
- [228] W. Rongben, G. Lie, T. Bingliang, and J. Lisheng, "Monitoring mouth movement for driver fatigue or distraction with one camera," in *Proc. 7th Int. IEEE Conf. Intell. Transp. Syst. (IEEE Cat. No 04TH8749)*, 2004, pp. 314–319.
- [229] M. Aljasim and R. Kashef, "E2DR: A deep learning ensemble-based driver distraction detection with recommendations model," *Sensors*, vol. 22, no. 5, 2022, Art. no. 1858.
- [230] G. Li, W. Yan, S. Li, X. Qu, W. Chu, and D. Cao, "A temporalspatial deep learning approach for driver distraction detection based on EEG signals," *IEEE Trans. Automat. Sci. Eng.*, early access, 2021, doi: 10.1109/TASE.2021.3088897.
- [231] J. Mafeni Mase, P. Chapman, G. P. Figueredo, and M. Torres Torres, "Benchmarking deep learning models for driver distraction detection," in *Proc. Int. Conf. Mach. Learn.*, Optim., Data Sci., 2020, pp. 103–117.
- [232] X. Wang, Y. Guo, J. Ban, Q. Xu, C. Bai, and S. Liu, "Driver emotion recognition of multiple-ECG feature fusion based on bp network and d-s evidence," *IET Intell. Transport Syst.*, vol. 14, no. 8, pp. 815–824, 2020.
- [233] C. Takano and Y. Ohta, "Heart rate measurement based on a time-lapse image," *Med. Eng. Phys.*, vol. 29, no. 8, pp. 853–857, 2007.
- [234] M. A. Hassan *et al.*, "Heart rate estimation using facial video: A review," *Biomed. Signal Process. Control*, vol. 38, pp. 346–360, 2017.

Vehicular Technology

- [235] S. Xu, L. Sun, and G. K. Rohde, "Robust efficient estimation of heart rate pulse from video," *Biomed. Opt. Exp.*, vol. 5, no. 4, pp. 1124–1135, 2014.
- [236] X. Chen, J. Cheng, R. Song, Y. Liu, R. Ward, and Z. J. Wang, "Video-based heart rate measurement: Recent advances and future prospects," *IEEE Trans. Instrum. Meas.*, vol. 68, no. 10, pp. 3600–3615, Oct. 2019.
- [237] R. Favilla, V. C. Zuccala, and G. Coppini, "Heart rate and heart rate variability from single-channel video and ICA integration of multiple signals," *IEEE J. Biomed. Health Informat.*, vol. 23, no. 6, pp. 2398–2408, Nov. 2019.
- [238] K. Alghoul, S. Alharthi, H. Al Osman, and A. El Saddik, "Heart rate variability extraction from videos signals: ICA vs. EVM comparison," *IEEE Access*, vol. 5, pp. 4711–4719, 2017.
- [239] M. Fukunishi, K. Kurita, S. Yamamoto, and N. Tsumura, "Noncontact video-based estimation of heart rate variability spectrogram from hemoglobin composition," *Artif. Life Robot.*, vol. 22, no. 4, pp. 457–463, 2017.
- [240] J. Xue *et al.*, "Driver's speech emotion recognition for smart cockpit based on a self-attention deep learning framework," in *Proc. 5th CAA Int. Conf. Veh. Control Intell.*, 2021, pp. 1–5.
- [241] A. F. Requardt, K. Ihme, M. Wilbrink, and A. Wendemuth, "Towards affect-aware vehicles for increasing safety and comfort: Recognising driver emotions from audio recordings in a realistic driving study," *IET Intell. Transport Syst.*, vol. 14, no. 10, pp. 1265–1277, 2020.
- [242] W. Li *et al.*, "A spontaneous driver emotion facial expression (defe) dataset for intelligent vehicles: Emotions triggered by video-audio clips in driving scenarios," *IEEE Trans. Affect. Comput.*, early access, 2021, doi: 10.1109/TAFFC.2021.3063387.
- [243] S. Shaees, H. Naeem, M. Arslan, M. R. Naeem, S. H. Ali, and H. Aldabbas, "Facial emotion recognition using transfer learning," in *Proc. Int. Conf. Comput. Inf. Technol.*, 2020, pp. 1–5.
- [244] R. K. Madupu, C. Kothapalli, V. Yarra, S. Harika, and C. Z. Basha, "Automatic human emotion recognition system using facial expressions with convolution neural network," in *Proc. 4th Int. Conf. Electron., Commun., Aerosp. Technol.*, 2020, pp. 1179–1183.
- [245] A. Poulose, J. H. Kim, and D. S. Han, "Feature vector extraction technique for facial emotion recognition using facial landmarks," in *Proc. Int. Conf. Inf. Commun. Technol. Convergence*, 2021, pp. 1072–1076.
- [246] M. Jeong, J. Nam, and B. C. Ko, "Lightweight multilayer random forests for monitoring driver emotional status," *IEEE Access*, vol. 8, pp. 60344–60354, 2020.
- [247] B. Verma and A. Choudhary, "A framework for driver emotion recognition using deep learning and grassmann manifolds," in *Proc. 21st Int. Conf. Intell. Transp. Syst.*, 2018, pp. 1421–1426.
- [248] S. B. Sukhavasi, S. B. Sukhavasi, K. Elleithy, A. El-Sayed, and A. Elleithy, "A hybrid model for driver emotion detection using feature fusion approach," *Int. J. Environ. Res. Public Health*, vol. 19, no. 5, 2022, Art. no. 3085.
- [249] J. Shi, C. Liu, C. T. Ishi, and H. Ishiguro, "Skeleton-based emotion recognition based on two-stream self-attention enhanced spatialtemporal graph convolutional network," *Sensors*, vol. 21, no. 1, 2020, Art. no. 205.
- [250] Wikipedia, "Remote keyless system," Accessed : Mar. 06, 2022. [Online]. Available: https://en.wikipedia.org/wiki/Remote\_keyless\_ system
- [251] X. Zeng, B. Wang, and K. J. R. Liu, "Driver arrival sensing for smart car using WiFi fine time measurements," in *Proc. Asia-Pacific Signal Inf. Process. Assoc. Annu. Summit Conf.*, 2020, pp. 41–45.
- [252] J. Steinberg, "Vulnerability in car keyless entry systems allows anyone to open and steal your vehicle," Accessed : Mar. 06, 2022. [Online]. Available: https://www.forbes.com/sites/josephsteinberg/2015/05/12/ vulnerability-in-car-keyless-entry-systems-allows-anyone-to-openand-steal-your-car/
- [253] M. Lake, "How it works; remote keyless entry: Staying a step ahead," Accessed : Mar. 06, 2022. [Online]. Available: https://www.nytimes.com/2001/06/07/technology/how-it-worksremote-keyless-entry-staying-a-step-ahead-of-car-thieves.html
- [254] P. H. Rettore, A. B. Campolina, A. Souza, G. Maia, L. A. Villas, and A. A. Loureiro, "Driver authentication in vanets based on intravehicular sensor data," in *Proc. IEEE Symp. Comput. Commun.*, 2018, pp. 00078–00083.
- [255] W. Yuan and Y. Tang, "The driver authentication device based on the characteristics of palmprint and palm vein," in *Proc. Int. Conf. Hand-Based Biometrics*, 2011, pp. 1–5.

- [256] I. Nakanishi, S. Baba, K. Ozaki, and S. Li, "Using brain waves as transparent biometrics for on-demand driver authentication," *Int. J. Biometrics*, vol. 5, no. 3–4, pp. 288–305, 2013.
- [257] S. Gupta, A. Buriro, and B. Crispo, "Driverauth: A risk-based multimodal biometric-based driver authentication scheme for ride-sharing platforms," *Comput. Secur.*, vol. 83, pp. 122–139, 2019.
- [258] Q. Xu, Y. Chen, B. Wang, and K. J. R. Liu, "Radio biometrics: Human recognition through a wall," *IEEE Trans. Inf. Forensics Secur.*, vol. 12, no. 5, pp. 1141–1155, May 2017.
- [259] J. Solaz et al., "Drowsiness detection based on the analysis of breathing rate obtained from real-time image recognition," *Transp. Res. Procedia*, vol. 14, pp. 3867–3876, Jan. 2016.
- [260] K. Lee, D. Han, and H. Ko, "Video analytic based health monitoring for driver in moving vehicle by extracting effective heart rate inducing features," *J. Adv. Transp.*, vol. 2018, Dec. 2018, Art. no. 8513487.
- [261] A. Pai, A. Veeraraghavan, and A. Sabharwal, "HRVCam: Robust camera-based measurement of heart rate variability," *J. Biomed. Opt.*, vol. 26, Feb. 2021, Art. no. 022707.
- [262] E. Nowara, T. Marks, H. Mansour, and A. Veeraraghavan, "SparsePPG: Towards driver monitoring using camera-based vital signs estimation in near-infrared," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit. Workshops*, 2018, pp. 1272–1281.
- [263] X. Wang, C. Yang, and S. Mao, "TensorBeat: Tensor decomposition for monitoring multi-person breathing beats with commodity wifi," *ACM Trans. Intell. Syst. Technol.*, vol. 9, pp. 1–27, Oct. 2017.
- [264] F. Adib, H. Mao, Z. Kabelac, D. Katabi, and R. C. Miller, "Smart homes that monitor breathing and heart rate," in *Proc. 33rd Annu. ACM Conf. Hum. Factors Comput. Syst.*, 2015, pp. 837–846.
- [265] Z. Yang, P. H. Pathak, Y. Zeng, X. Liran, and P. Mohapatra, "Monitoring vital signs using millimeter wave," in *Proc. 17th ACM Int. Symp. Mobile Ad Hoc Netw. Comput.*, 2016, pp. 211–220.
- [266] M. Mercuri, I. R. Lorato, Y. Liu, F. Wieringa, C. V. Hoof, and T. Torfs, "Vital-sign monitoring and spatial tracking of multiple people using a contactless radar-based sensor," *Nature Electron.*, vol. 2, pp. 252–262, Jun. 2019.
- [267] M. Zhao, F. Adib, and D. Katabi, "Emotion recognition using wireless signals," in *Proc. 22nd Annu. Int. Conf. Mobile Comput. Netw.*, 2016, pp. 95–108.
- [268] V. L. Petrovic, M. M. Jankovic, A. V. Lupsic, V. R. Mihajlovic, and J. S. Popovic-Bozovic, "High-accuracy real-time monitoring of heart rate variability using 24 GHz continuous-wave doppler radar," *IEEE Access*, vol. 7, pp. 74721–74733, 2019.
- [269] N. T. P. Nguyen, P.-Y. Lyu, M. H. Lin, C.-C. Chang, and S.-F. Chang, "A short-time autocorrelation method for noncontact detection of heart rate variability using CW doppler radar," in *Proc. IEEE MTT-S Int. Microw. Biomed. Conf.*, 2019, vol. 1, pp. 1–4.
- [270] T. Ohtsuki and E. Mogi, "Heartbeat detection with doppler radar based on estimation of average R-R interval using viterbi algorithm," in *Proc. IEEE 27th Annu. Int. Symp. Pers., Indoor, Mobile Radio Commun.*, 2016, pp. 1–5.
- [271] F. Wang, X. Zeng, C. Wu, B. Wang, and K. J. R. Liu, "Driver vital signs monitoring using millimeter wave radio," *IEEE Internet Things J.*, early access, 2021, doi: 10.1109/JIOT.2021.3128548.
- [272] G. Vinci, T. Lenhard, C. Will, and A. Koelpin, "Microwave interferometer radar-based vital sign detection for driver monitoring syst," in *Proc. IEEE MTT-S Int. Conf. Microw. Intell. Mobility*, 2015, pp. 1–4.
- [273] F. Wang, X. Zeng, C. Wu, B. Wang, and K. J. R. Liu, "mmHRV: Contactless heart rate variability monitoring using millimeter-wave radio," *IEEE Internet Things J.*, vol. 8, no. 22, pp. 16623–16636, Nov. 2021.
- [274] R. Fletcher and J. Han, "Low-cost differential front-end for doppler radar vital sign monitoring," in *Proc. IEEE MTT-S Int. Microw. Symp. Dig.*, 2009, pp. 1325–1328.
- [275] E. Schires, P. Georgiou, and T. S. Lande, "Vital sign monitoring through the back using an UWB impulse radar with body coupled antennas," *IEEE Trans. Biomed. Circuits Syst.*, vol. 12, no. 2, pp. 292–302, Apr. 2018.
- [276] C. Lin, R. Wu, S. Liang, W. Chao, Y. Chen, and T. Jung, "EEG-based drowsiness estimation for safety driving using independent component analysis," *IEEE Trans. Circuits Syst. I: Regular Papers*, vol. 52, no. 12, pp. 2726–2738, Dec. 2005.
- [277] F. Zhang, C. Wu, B. Wang, H. Lai, Y. Han, and K. J. R. Liu, "WiDetect: Robust motion detection with a statistical electromagnetic model," *Proc. ACM Interact. Mobile Wearable Ubiquitous Technol.*, vol. 3, pp. 1–24, Sep. 2019.

- [278] S. D. Regani, B. Wang, and K. J. R. Liu, "WiFi-based device-free gesture recognition through-the-wall," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process.*, 2021, pp. 8017–8021.
- [279] F. Zhang, C. Chen, B. Wang, and K. J. R. Liu, "WiSpeed: A statistical electromagnetic approach for device-free indoor speed estimation," *IEEE Internet Things J.*, vol. 5, no. 3, pp. 2163–2177, Jun. 2018.
- [280] B. Wang, Q. Xu, C. Chen, F. Zhang, and K. J. R. Liu, "The promise of radio analytics: A future paradigm of wireless positioning, tracking, and sensing," *IEEE Signal Process. Mag.*, vol. 35, no. 3, pp. 59–80, May 2018.
- [281] K. J. R. Liu and B. Wang, Wireless AI: Wireless Sensing, Positioning, IoT, and Communications. Cambridge, U.K.: Cambridge Univ. Press, 2019.
- [282] F. Wang, F. Zhang, C. Wu, B. Wang, and K. J. R. Liu, "Passive people counting using commodity WiFi," in *Proc. IEEE 6th World Forum Internet Things*, 2020, pp. 1–6.
- [283] F. Wang, X. Zeng, C. Wu, B. Wang, and K. J. R. Liu, "Radio frequency based heart rate variability monitoring," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process.*, 2021, pp. 8007–8011.
- [284] F. Wang, F. Zhang, C. Wu, B. Wang, and K. J. R. Liu, "Vimo: Vital sign monitoring using commodity millimeter wave radio," in *Proc. IEEE Int. Conf. Acoust., Speech, Signal Process.*, 2020, pp. 8304–8308.
- [285] A. Guettas, S. Ayad, and O. Kazar, "Driver state monitoring system: A review," in *Proc. 4th Int. Conf. Big Data Internet Things*, 2019, pp. 1–7.
- [286] M. Verleysen and D. François, "The curse of dimensionality in data mining and time series prediction," in *Proc. Int. Work- Conf. Artif. Neural Netw.*, 2005, pp. 758–770.



**XIAOLU ZENG** received the B.S. degree from the Harbin Institute of Technology, Harbin, China, in 2014, and the Ph.D. degree from the School of Electronic Engineering, Xidian University, Xi'an, China, in 2020. From July 2020 to December 2021, he was a Postdoc Research Associate, and from September 2017 to June 2020 a Visiting Student, with the Department of Electrical and Computer Engineering, University of Maryland, College Park, MD, USA. He is currently an Assistant Professor with the School of Information

and Electronics, Beijing Institute of Technology, Beijing, China. His research interests include intelligent wireless sensing, Internet of Things, and advanced driver assistance systems.



**FENGYU WANG** (Member, IEEE) received the B.S. and M.S. degrees from the School of Information and Communication Engineering, Beijing University of Posts and Telecommunications, Beijing, China, in 2014 and 2017, respectively, and the Ph.D. degree in electrical engineering from the University of Maryland, College Park, MD, USA, in 2021. She is currently a Lecturer with the School of Artificial Intelligence, Beijing University of Posts and Telecommunications, Beijing, China. Her current research interests include Inter-

net of Things, wireless sensing, statistical signal processing, and semantic communications.



**BEIBEI WANG** (Senior Member, IEEE) received the B.S. degree (Hons.) in electrical engineering from the University of Science and Technology of China, Hefei, China, in 2004, and the Ph.D. degree in electrical engineering from the University of Maryland, College Park, MD, USA, in 2009. She was with the University of Maryland as a research associate during 2009–2010, and with Qualcomm Research and Development during 2010–2014. Since 2015, she has been with Origin Wireless Inc., where she is currently the Vice

President of Research. She is a coauthor of *Cognitive Radio Networking* and Security: A Game-Theoretic View (Cambridge University Press, 2010) and Wireless AI: Wireless Sensing, Positioning, IoT, and Communications (Cambridge University Press, 2019). Her research interests include Internet of Things, mobile computing, wireless sensing and positioning. Dr. Wang was the recipient of the 2020 IEEE Internet of Things Journal Best Paper Award, 2015 IEEE Signal Processing Society Overview Paper Award, and several research and invention awards from the University of Maryland. She was on the Editorial Board of the IEEE SIGNAL PROCESSING LETTERS, IEEE INTERNET OF THINGS JOURNAL, and IEEE JOURNAL ON SELECTED AREAS IN COMMUNICATIONS.



**CHENSHU WU** (Senior Member, IEEE) received the B.E. degree from the School of Software, Tsinghua University, Beijing, China, in 2010, and the Ph.D. degree in computer science from Tsinghua University, in 2015. He is currently an Assistant Professor with the Department of Computer Science, The University of Hong Kong, Hong Kong. He is also the Chief Scientist with Origin Wireless Inc. His research interests include wireless AIoT systems at the intersection of wireless sensing, ubiquitous computing, digital health, and the Inter-

net of Things.



**K. J. RAY LIU** (Fellow, IEEE) is the founder and CEO of Origin AI that pioneers AI for wireless sensing and indoor tracking. The invention of wireless AI won three prestigious CES Innovation Awards, including CES Best of Innovation in 2021.

He was Distinguished University Professor and Christine Kim Eminent Professor of information technology of the University of Maryland, College Park, from where he retired after over three decades of career in education. His research contri-

butions encompass broad aspects of signal processing and communications. He has trained more than 70 doctoral/postdoctoral students, of which 10 are now IEEE fellows.

Dr. Liu is the recipient of the 2021 IEEE Fourier Technical Field Award for Signal Processing, 2016 IEEE Leon K. Kirchmayer Graduate Teaching Award, IEEE Signal Processing Society 2014 Society Award, and IEEE Signal Processing Society 2009 Technical Achievement Award. Recognized as a Web of Science Highly Cited Researcher, he is a Fellow of the American Association for the Advancement of Science (AAAS), and the U.S. National Academy of Inventors.

Dr. Liu is 2022 IEEE President and CEO. He was the 2019 IEEE Vice President for Technical Activities. He was also the 2012-2013 President of IEEE Signal Processing Society, where he once served as the Editor-in-Chief of *IEEE Signal Processing Magazine*.



**OSCAR C. AU** (Fellow, IEEE) received the B.A.Sc. degree from the University of Toronto, Toronto, ON, Canada, in 1986, and the M.A. and Ph.D. degrees from Princeton University, Princeton, NJ, USA, in 1988 and 1991 respectively. After one year of Postdoc, he joined the Hong Kong University of Science and Technology, Hong Kong, as an Assistant Professor. Associate Professor, and eventually a Full Professor. He did research in signal processing, statistical communication, optimization, and machine learning with more than

460 publications in top journals and conferences, more than 80 standard contributions, and more than 30 patents.

Dr. Au an HKIE Fellow. He chaired several international signal processingrelated technical committees in IEEE and APSIPA, and was a member in several more. He was an Associate Editor for more than 10 international journals, including some top IEEE journals in respective fields. He also chaired several international conferences and served on the organizing committee of some other. In 2016, he joined Origin Wireless as the Vice President and worked on wireless sensing.